

Received 10 August 2024, accepted 3 September 2024, date of publication 6 September 2024, date of current version 25 September 2024. Digital Object Identifier 10.1109/ACCESS.2024.3455900

SURVEY

Patient Monitoring System Based on Internet of Things: A Review and Related Challenges With Open Research Issues

BASHEERA M. MAHMMOD^{®1}, MARWAH ABDULRAZZAQ NASER¹, AHLAM H. SHANIN AL-SUDANI¹, MUNTADHER ALSABAH², HALA J. MOHAMMED¹, HAYA ALASKAR^{®3}, FAHDAH ALMARSHAD^{®4}, ABIR HUSSAIN^{®5,6}, (Senior Member, IEEE), AND SADIQ H. ABDULHUSSAIN^{®1}

¹Department of Computer Engineering, University of Baghdad, Baghdad 10071, Iraq

²Medical Technical College, Al-Farahidi University, Baghdad 10071, Iraq

³Department of Computer Sciences, College of Computer Engineering and Sciences, Prince Sattam Bin Abdulaziz University, Al-Kharj 16278, Saudi Arabia

⁴Department of Information Systems, College of Computer Engineering and Sciences, Prince Sattam Bin Abdulaziz University, Al-Kharj 16278, Saudi Arabia

⁵Department of Electrical Engineering, University of Sharjah, Sharjah, United Arab Emirates

⁶School of Computer Science and Mathematics, Liverpool John Moores University, L3 3AF Liverpool, U.K.

Corresponding author: Haya Alaskar (h.alaskar@psau.edu.sa)

This paper is partially funded from Prince Sattam bin Abdulaziz University project number (PSAU/2024/R/1445).

ABSTRACT The unexpectedly high number of deaths caused by inadequate medical care is, to date, considered as a serious problem. Moreover, the ratio of elderly people who require continuous care is rising. Therefore, a patient monitoring system (PMS) also known as remote patient monitoring (RPM) using the latest Internet of Things (IoT) technology becomes a viable solution that can provide efficient healthcare from a remote distance. PMS monitors timely physiological signals of a patient's health and can reduce the healthcare costs of treatment significantly. In PMS, different health and vital signs issues such as body temperature, heart rate, sleep monitoring, fall detection, and blood pressure can be checked effectively in real-time. To this end, this paper provides a clear vision of electronic healthcare assistance based on PMS and explores the applications of IoT that allow efficient medical services in healthcare systems. In particular, the objective of this paper is to provide a review of PMS, current research, and the challenges associated with this area. Besides, the essential services that can be offered by PMS for monitoring human activities are also discussed. Furthermore, the communication networks and protocols that are required to endure efficient healthcare systems are explained. Finally, this paper discusses several research challenges and open issues that can be investigated for further work. Overall, this paper offers valuable insights for both industry professionals and academic researchers, exploring potential avenues for new research directions.

INDEX TERMS Edge computing, healthcare, IoT, microcontrollers, patient monitoring system (PMS), remote diagnosis, remote monitoring, sensors, wearable devices.

I. INTRODUCTION

The Internet of Things (IoT) is utilized in many essential fields and has become one of the most important technologies worldwide. IoT is a revolutionary technology, which is able to connect multiple devices, collect heterogeneous information/data from multiple sources and devices, and

The associate editor coordinating the review of this manuscript and approving it for publication was Tao $\text{Liu}^{(D)}$.

transmit and process this information and data in realtime. IoT is defined as a network of things, comprising different physical things, encompassing embedded devices integrated with various software and technologies [1], [2]. IoT allows different types and sizes of things such as cameras, vehicles, phones, home appliances, buildings, industrial systems, and people to communicate with each other and share information in real-time, hence, achieving intelligent systems [3]. The primary function of IoT revolves

© 2024 The Authors. This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License.

For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

around facilitating communication and data transmission via the Internet, thereby enabling seamless connectivity and interaction between these interconnected physical things. In IoT, the term "Internet" denotes a worldwide network comprising billions of computers and diverse electronic devices. Through the Internet, users can access the desired information and connect with other people from anywhere and at any time through standard protocols. On the other hand, the term "things" refers to any object that can be associated with a particular connectivity. IoT is used to connect anything and get the desired service [4]. Besides, the escalation in data transmission and repository size has increased the demand for big data transfer, so the IoT has become very essential in data exchange. IoT represents a groundbreaking paradigm in which objects possessing distinct identities can seamlessly integrate into an interconnected information network to provide intelligent services [5].

In recent years, there have been considerable advancements in sensing technologies, wearable/implantable devices, and wireless communication technologies. The development of network technology and sensors such as laser sensors, radar sensors, and camera devices, has allowed a smart environment to be achieved with homogeneous and heterogeneous devices. IoT networks leverage a range of monitoring systems that harness the benefits of data processing and analytics, utilizing the power of the Internet to enable informed decision-making in real-time [6]. IoT networks have evolved in many applications such as patient monitoring system (PMS) [7], [8], [9], intelligent energy management [10], smart city [11], smart home monitoring [12], [13], industrial application [14], vehicle monitoring system [15]. The integration of IoT applications and engineering technologies has demonstrated significant value in the biomedical domain, enhancing the efficiency of medical practitioners and optimizing treatment processes. The convergence of IoT with medical devices has expedited medical interventions, streamlined diagnostic procedures, and improved the management of chronic illnesses. From the technological perspective, IoT involves some popular technologies such as wireless-body-area-networks (WBANs) to transfer the collected information from the devices to the centralized cloud for analyzing and extracting meaningful information for efficient decision-making.

The rapid growth of IoT devices brings innumerable advantages. According to Statista, there will be more than 75 billion IoT-devices connected to the Internet by 2025, resulting in an enormous data output [16]. Moreover, IoT is growing on an everyday basis with many potential innovative technologies [17]. To this end, IoT is expected to have a potential economic impact of \$3-6 trillion, and the largest fraction, \$1-2.5 trillion, of this economic impact comes from smart healthcare applications [18].

The projected global market size for IoT in healthcare was valued at USD 180 billion in 2022 and is anticipated to grow to approximately USD 962.21 billion by 2032, reflecting a compound annual growth rate (CAGR) of 18.3%



FIGURE 1. The estimated IoT in healthcare market sizes, 2022 to 2032 (USD Billion) [19].

from 2023 to 2032, as shown in FIGURE 1 [19]. The predicted market share of VR/AR by 2021 is 54.6 billion USD, up from 2.5 billion USD in 2016 [17]. Consequently, the bandwidth consumption by this industry is also increased. The connected devices in different healthcare institutes and the utilization of IoT systems are considered the most important parameters related to market growth.

Therefore, leveraging IoT within healthcare is highly advantageous. IoT-driven healthcare minimizes human error by networking all vital sign monitoring devices to a decision support system, empowering physicians to deliver more precise and timely diagnoses [20]. One of the most important applications of IoT in the healthcare field is PMS [21], [22]. The Internet of Medical Things (IoMT) represents the convergence of medical devices with the Internet of Things (IoT) [23]. IoMT leads the future of healthcare systems, envisaging a scenario where every medical device is interconnected and remotely monitored by healthcare professionals via the Internet [24]. In the medical sector, the importance of IoT technologies is highlighted by IoMT, which is a collection of medical and fitness devices with various applications that are connected to the systems of healthcare through online computer networks. IoMT has several advantages such as improving drug management, decreasing healthcare costs, enhancing patient experience, improving diagnosis and treatment, improving disease management and most effectively achieving efficient PMS for chronic diseases [25]. This paradigm shift promises faster and more cost-effective healthcare delivery as it continues to evolve. Hence, PMS is considered a major part of IoT, which is related specifically to the concept of the IoMT [21]. In general, the network of IoMT is dedicated to various medical systems such as PMS, disease treatment, anomaly detection, medical nursing and rehabilitation, remote and telemedicine care, conditioning, and screening systems. PMS combines different equipment that constantly monitors patients' health through warning systems of patient vital signs based on the detecting and tracking of the changes in their health condition [26]. The importance of PMS has rapidly increased due to the highly contagious diseases (such as COVID-19) and the increasing number of elderly people who require constant care where the demographic shift toward the aging population [27]. A PMS is useful for all patients suffering from serious illnesses such as diabetes,

cardiovascular disease, mental illness, cancer, hypertension, and COVID-19 [28], [29]. In addition, there is a growing incidence of factors that negatively impact the quality of life, such as busy schedules, outbreaks of epidemic diseases, and increasing pollution levels. Recent statistics indicate that over 90% of the population is exposed to polluted environments, exacerbating these challenges [30]. Therefore, PMS appears as an important technology, which enables the medical professionals to monitor the patient's health remotely and quickly in addition to revolving the healthcare systems with lower costs and better patient outcomes. The basic functionalities of PMS are signs detection, monitoring, and tracking. The application of PMS allows specialists to monitor and track patients remotely in a real-time manner based on their convenience at home or office. Besides, PMS allows to warn the caregivers of potential threats to patient health, and hence to take quick action automatically. There are various detection, monitoring, and tracking methods that have been developed for PMS [31].

Generally, the applications of PMS have analogous architecture, which consists of wearable sensors and microcontrollers, wireless communication networks, and cloud computing platforms. The work in [32] and [33] demonstrated that PMS can be implanted in real-time with multi-tier pervasive WBAN. There are many types of sensors that are used to detect vital signs and physiological conditions of the patient. Examples of these physiological parameters are heart rate, body temperature, electrocardiogram (ECG), electroglottography (EGG), electroencephalogram (EEG), blood oxygen saturation level (SpO2), blood pressure, electromyogram (EMG), electrooculogram (EOG), magnetoencephalogram (MEG), breathing rate, mechanomyogram (MMG), photoplethysmogram (PPG), respiration (RESP), and electrodermal activity (EDA) [23], [34]. Sensors send data to the microcontrollers, which then analyze the data and determine the process based on the algorithmic design implemented. For instance, if a temperature body sensor detects a change in the body temperature, the sign will be sent remotely to the specialist to sound an alarm and apply a quick treatment to the patient by the specialist. Hence, efficient communication systems that are able to carry out fast detection to send this essential information in a reliable and secure way are required.

To this end, wireless communication technology has advanced significantly in recent years, which is considered a crucial part of the development of automated and smart healthcare tasks [35], [36], [37]. This advanced development in such technologies allows PMS to provide fast detection of patient conditions in real-time scenarios. Therefore, PMS continues to serve as a cornerstone in driving the evolution of autonomous healthcare services, particularly as wireless communication technology advances. Its ubiquity ensures widespread accessibility to healthcare while minimizing costs and errors, thereby contributing significantly to the advancement of medical care delivery. Moreover, through the employing of efficient cloud computing for PMS, we can get coherent-optimized healthcare systems. Utilizing cloud computing offers several advantages for healthcare systems. These advantages include streamlined processing of healthrelated issues, simplified management of various diseases, and the capability to access real-time patient information seamlessly and securely. Furthermore, cloud-based systems enable synchronized data sharing across multiple platforms, ensuring scalability to accommodate varying workloads while addressing concerns regarding scalability and security. Besides, employing cloud computing with robust algorithms could ensure that critical information remains readily accessible whenever needed, thereby enhancing the efficiency and effectiveness of healthcare delivery. A medical server refers to a remote computer situated within a healthcare institution, tasked with real-time data monitoring and offering health recommendations to patients. Physicians or a database handle monitoring and post-processing. The medical server plays a crucial role in remotely monitoring patients in telemedicine settings, using a remote computer to collect and transmit vital signs to a telemedicine server for analysis. Subsequently, the medical server assists healthcare professionals by recommending suitable healthcare services for patients remotely based on the analyzed vital signs

A. MOTIVATION AND PAPER CONTRIBUTIONS

During epidemics and disease crises, such as the COVID-19 pandemic, and for individuals managing chronic health conditions like diabetes, as well as for the elderly population, PMS can play a crucial role in addressing the shortcomings of traditional healthcare models. PMS becomes crucial, especially in regions with limited access to healthcare facilities and medical professionals. PMS can facilitate the efficient transmission of patient data to healthcare providers, enabling timely interventions and necessary actions based on real-time information. Given these critical needs and the challenges faced by conventional healthcare systems, there is a compelling necessity to comprehensively study and understand PMS. Noting that, the appropriate PMS architecture and necessary communication technologies were not identified in the earlier review studies. This paper aims to provide a literature review and an in-depth study of recent trends in PMS based on IoT technologies, which are required to achieve efficient healthcare services.

The contribution of this manuscript can be summarized as follows:

- We provide a comprehensive review of the architecture of PMS, discussing its essential technologies and components. This review serves as a foundation for understanding the complexities of PMS and its potential applications in healthcare systems.
- We highlight the importance and urgency of adopting PMS-based IoT in healthcare systems, emphasizing its potential to revolutionize healthcare delivery and improve patient outcomes. By adopting PMS-based IoT, healthcare systems can become more efficient, effective, and patient-centered.

- We analyze various types of existing monitoring and tracking methods in healthcare systems, providing a thorough understanding of the current state-of-theart approaches. This analysis serves as a basis for identifying areas of improvement and opportunities for innovation.
- We present the relationship between different layers in the architecture of PMS, providing a detailed understanding of how these layers interact and interoperate. This understanding is essential for designing and implementing effective PMS-based systems.
- We identify challenges and provide recommendations for future work, benefiting both academic and industrial sectors in remote healthcare applications. These challenges and recommendations serve as a roadmap for future research and development in the field.
- Finally, this survey paper contributes to improving healthcare delivery, disease surveillance, management, and response strategies during public health emergencies and routine care scenarios. By leveraging the potential of PMS-based IoT, healthcare systems can become more resilient, responsive, and effective in addressing the needs of patients and communities.

B. PAPER ORGANIZATION

The structure of this manuscript is outlined as follows. Section II presents the basic terminologies related to PMS and provides introductory concepts of PMS in healthcare systems. Section III provides a review of remote patient monitoring systems and discusses the architecture used for PMS based on IoT, which is essentially required to achieve effective healthcare systems. Section IV provides a thorough explanation of the sensors' application in PMS. Section V gives a detailed description of the communication technologies required for PMS. Section VI highlights the cloud computing and distribution technologies and their applications in the PMS. Section VIII provides research challenges and discusses the future research direction in PMS based on IoT in healthcare systems. Finally, this paper is concluded in Section IX. FIGURE 2 shows each section and subsection in the paper.

II. BASIC TERMINOLOGIES AND INTRODUCTORY CONCEPTS OF PMS IN HEALTHCARE SYSTEMS

This section discusses the essential terminologies that are commonly used in healthcare monitoring. In addition, the basic concept of PMS systems is introduced.

A. SOME BASIC TERMINOLOGIES IN HEALTH CARING DOMAIN

Over the last few years, the advancement in health technologies has demonstrated a considerable consequence on our quality of life. In particular, the development of these advanced technologies has a considerable impact on enhancing patient care and diagnostics, which helps in providing better and quicker treatment. In conventional healthcare systems, doctors need to visit patients physically and required the use of traditional tools for diagnosing their cases [38]. The traditional methods are considered to be cost-effective, leading to delayed diagnosis, especially for remote distances, which leads to serious consequences. Therefore, one of the most important goals of advanced technologies is to build smart and efficient healthcare models that are able to close the gap between caregivers and patients to improve patients' well being [39]. Specifically, with the advancements of mobile technologies and smart medical devices and sensors, health experts become able to make a great improvement on healthcare systems. Furthermore, m-health and e-health, supported by Information and Communication Technology (ICT), enable the efficient delivery of healthcare services to multiple patients, aiding in their health improvement. Besides, the emergence of IoT presents even more advantages by connecting diverse devices to the Internet, thereby providing caregivers with real-time updates on patients' conditions [7]. In the epicenter of advanced healthcare technologies, PMS is considered in the advancement of a medical stream. PMS has different types of equipment such as microcontrollers and sensors, which are used to collect the information of the patients and give the required process using IoT. PMS is very helpful in remote monitoring patients, especially in the conditions of elderly and chronically ill patients and those who have chronic diseases such as heart disease, diabetes, hypertension, and pressure. This remote monitoring capability offered by PMS proves particularly beneficial for ensuring timely intervention and personalized care for elderly individuals and those managing chronic conditions. By enabling continuous monitoring and early detection of health issues, PMS plays a pivotal role in improving the quality of life for patients, allowing healthcare providers to intervene proactively and mitigate potential complications. The following are some basic terminologies that are used for PMS.

- Conventional healthcare system: This refers to a traditional healthcare practices, where a doctor visits patients with the required traditional tools, that are not essentially related to information and communication technology (ICT) [38].
- Medical emergency: This refers to the manual call to the hospital in emergency situations. This call should constitute some critical information such as the location, the medical problem nature, and a valid available contact until the ambulance arrives. Once the patient reaches the hospital, the caregiver records the vital parameters such as blood pressure, breath rate, heartbeat, etc., to proceed with the required treatment.
- The Internet of Medical Things (IoMT), also known as the Internet of Health Things (IoHT), represents a significant advancement in healthcare, leveraging IoT technologies to enhance patient care and healthcare services [8]. IoMT encompasses various medical devices connected to the Internet, including wearable sensors and implantable devices like smartwatches



FIGURE 2. Structure of the PMS review paper.

and smartphones, facilitating the analysis of patient data [39], [40].

• Telemedicine: This term is defined as an integrated system that has bidirectional remote contact among

patients and medical professionals. Telemedicine utilizes the emergence of ICT to provide clinical healthcare and to exchange health information in remote locations. This strategy is especially beneficial for remote areas with limited access to healthcare services, as it overcomes geographical barriers and enhances medical care accessibility. Telemedicine is crucial for delivering urgent care during emergencies, potentially saving lives in critical scenarios. Additionally, telemedicine plays a role in monitoring patients at home, facilitating quicker, more efficient, and cost-effective patient mobilization and rehabilitation [41].

- Remote Patient Monitoring system (RPM): This term is a standard for healthcare delivery that enables caregivers to monitor patients remotely using smart body sensors and build upon modern connectivity standards [8]. RPM performs certain specific tests on the patient's body, which is essential in healthcare, especially for the elderly and chronically ill. PMS term is similar to Remote Health Monitoring System (RHMS), Mobile Health Monitoring System (MHMS), and Wearable Health Monitoring System (WHMS), each denoting similar systems that serve similar purposes just with different names across literature [22].
- Sensors and wearable devices: This refers to technologyenabled tools that are able to monitor physiological parameters and activity levels for many patients, facilitating continuous health monitoring and tracking and personalized care interventions. Typically, sensors receive their input signals from the physical surroundings and produce a response action based on the received signals. After the vital signs or signals are received by sensors, they transmit them wirelessly to the Body Area Networks (BAN) control unit. By monitoring vital signs like body temperature, blood pressure, body temperature, serum cholesterol, glucose level, arterial oxygen saturation, and breath rate, wearable sensors provide enormous potential for the early detection of diseases. Noting that wearable devices are typically attached to the human body, or they are integrated into elastic bands, textile fiber, or patient clothes. Wearable devices are used to measure the physiological signals of the patients and their activity. Moreover, wireless medical sensors are used to transmit vital signs to a remote server for diagnostic purposes.
- Smart health system: This concept has been given multiple definitions with different interpretations [39]. Smart health can be defined as an integration of advanced technologies, data analytics, and digital communication tools to optimize health management, improve medical service, enhance patient outcomes, and improve healthcare delivery.
- Pervasive health: This concept aims to deliver healthcare services to patients anytime and anywhere, as reducing institutionalization being a tool to face healthcare costs



FIGURE 3. A generic PMS system with BSN, local switch, PMS server, and dashboard for PMS visualization with medical staff.

while increasing both the coverage and the quality of healthcare [39]. Based on the given definition, the concept of pervasiveness emphasizes the broader societal impact of healthcare accessibility to all, rather than solely focusing on technological elements. Pervasive healthcare also encompasses the usage of pervasive computing principles, such as IoT, to deliver medical services directly to homes or anywhere. Additionally, it involves remote data collection through mobile devices and sensor networks, often resulting in large volumes of data in diverse formats and with high frequency.

• Vital patients' signs: The human body contains different health signs that reflect the critical health status and can be effectively used to help in disease detection. These signs are defined as the important indicators that are used to monitor the condition of a patient by allowing the specialized to know the body functions and find if there is any delay in the patient's recovery. Checking these patient signs is one of the most common interventions being practiced in healthcare settings. The most common types of these signs are heart rate, body temperature, ECG, EGG, EEG, SpO2, blood pressure, EMG, EOG, MEG, breathing rate, MMG, PPG, RESP, and EDA [23], [34].

Health monitoring based on IoT is considered as an advanced platform to provide us with a remarkable service in contemporary medicine. Recent advances in cloud computing, mobile technologies, and wireless sensor networking give a great possibility for efficient remote patient monitoring. FIGURE 3 demonstrates a generic PMS system

with body-sensor-network (BSN), local switch, PMS server, and dashboard for PMS visualization with medical staff. The applications of IoT in the medical sector seem to be endless, but the most promising ones are represented by monitoring and tracking patient health using PMS. To this end, PMS is able to perform various functions with an aim to achieve different objectives, which can be listed as follows:

- Affordability and automating healthcare: Affordability in healthcare can be significantly enhanced by PMS through measures such as reducing hospital readmissions and administrative costs, ultimately lowering overall healthcare expenses. Additionally, PMS contributes to automating healthcare processes, optimizing tasks such as data collection, appointment scheduling, and treatment reminders, thereby improving efficiency and quality of care delivery.
- Improve work productivity and accuracy: Monitoring patients through PMS will likely improve the accuracy of the diagnosis and provide a fast decision in medication management and productivity [42]. PMS has the ability to simplify healthcare workflows and reduce administrative burdens on healthcare professionals, hence allowing them to focus more on patient care and improve overall work productivity.
- Real-time patient tracking: The majority of PMS provides portable, high reliability, and wireless low-cost tracking of patient health metrics and activities, hence facilitating prompt decision-making and interventions based on current patient conditions. In the sophisticated technologies of real-time patient location monitoring, the location of the patients is tracked then an alarm is raised based on specific incidents. Such activity should be performed in real-time, and hence, effective rescue and immediate operations should be guaranteed.
- Security of data access: PMS can offer the security and reliability requirements for healthcare systems, which is important to protect the data and prevent unauthorized access by a secure communication framework. Besides, PMS ensures the privacy and confidentiality of patient information by using robust encryption and access control measures, safeguarding sensitive medical information from unauthorized access, and following industry standards and regulations. The main criteria for security and privacy of a cloud-based healthcare approach are authorization, non-repudiation, authentication, integrity, and confidentiality.
- Speed up data management: in PMS, patient's vital parameters can be transmitted faster in real-time to health professionals such as doctors, nurses, or laboratories regardless of the geographic location of patients. Therefore, the rapid availability of patient information allows rapid data analysis and provides predictive insights into the patient's condition with appropriate management and treatment. In addition, PMS accelerates the processing and organization of



FIGURE 4. The significant functions and applications of PMS in healthcare systems.

patient data, allowing healthcare providers to access critical information efficiently.

- Remote monitoring and medical assistance: One of the main objective of PMS is to allow healthcare professionals to monitor patients' vital signs remotely in real-time, hence enabling timely intervention and medical assistance, especially for those in remote or inaccessible areas.
- Prevention of disease spread: By enabling remote monitoring and minimizing in-person interactions, PMS helps prevent the spread of infectious diseases, especially during pandemics or outbreaks, by reducing the need for patients to visit hospitals and doctors.
- Reduction of care costs and patient expenses: PMS can play an essential role in lowering healthcare costs. This can be achieved by detecting health issues in an early stage, proactive management, and reduced hospitalizations, and hence, beneficial for both providers and patients, resulting in decreased overall expenses [43].
- Patient self-management: PMS enables patients to participate in their own healthcare by providing tools and resources for self-monitoring and management of their health conditions. By promoting self-management, healthcare providers can enhance patient engagement, improve health outcomes, and optimize the efficient use of healthcare resources.
- Boosts caregiver connectivity: PMS enhances communication and collaboration among caregivers, enabling seamless coordination of patient care plans, sharing of medical information, and providing support to patients and their families.

The aforementioned functions and applications are illustrated in FIGURE 4.

III. REMOTE PATIENT MONITORING SYSTEMS ARCHITECTURE BASED ON IOT

Currently, one of the most critical global challenges revolves around ensuring that healthcare services are available, accessible, and affordable for everyone. PMS is, in particular, one of the most essential health systems that has a great effect in health services by providing faster monitoring, diagnosis, and treatment for people who need continuous and/or urgent healthcare. PMS is a revolution in the digitalization of healthcare, and it uses a combination of software and hardware to provide a fast and effective decision-making process. We aim to build a healthcare system that are able to allow individuals to take an active role in the management of their health conditions, which can effectively improve patient engagement and enhance overall health outcomes. This engagement impacts individuals and communities profoundly, influencing health outcomes and overall well-being. Thus, by enabling people to actively engage in their healthcare decisions and treatment plans, patient self-management plays a critical role in tackling health concerns. By fostering patient engagement and responsibility, healthcare providers can enhance the effectiveness of care delivery and the distribution of healthcare services, leading to improved health outcomes and a more sustainable healthcare system [44].

One of the main objectives of using IoT and remote health monitoring is to transfer medical services from hospitals and extend them to convenient homes. IoT comprises a network of interconnected electrical and wireless devices that collaborate to collect and share patient data for diagnostic and analytical purposes, as well as for securely storing patient information. This technology has revolutionized healthcare by allowing real-time health monitoring, and personalized treatments that could significantly improve patient healthcare [45]. Various sensors and devices can be used in PMS to collect patient data. PMS-based IoT is an advancement in the medical stream for monitoring patients' conditions. For instance, the movement of comma patients' bodies should be recognized and checked continuously using temperature measuring devices, IoTs accelerometer (devices and sensors for measuring body movement), and eye blinkers [22]. The elderly and people with chronic illnesses can benefit from these remote monitoring devices [46].

The design and implementation of PMS architectures aim to optimize patient care by ensuring accurate data collection, efficient data processing, and timely communication of relevant information to healthcare providers. To this end, several research works have considered the real-world applications of PMS based on IoT in healthcare systems. For example, the work in [47] provided a comprehensive survey discussing edge computing solutions for IoT applications, covering applications and highlighted the benefits of edge computing over cloud computing in healthcare domains. The work in [48] investigated the impact of IoT intervention in a hospital unit and provided empirical evidence on the effects

VOLUME 12, 2024

of smart technologies on patient safety particularly (patient falls and hand hygiene compliance rate) and staff experiences. The work in [49] considered the detection and monitoring of cardiovascular autonomic neuropathy in diabetes patients. The work in [50] provided an empirical investigation of how wearable IoT devices would bring about a revolution in the healthcare industry. The work in [51] developed a system that tracks blood pressure using the Keep In Touch (KIT) method and integrated closed-loop healthcare services. The KIT device is connected to a JAVA-based mobile phone via near-field communication, which operates on principles of magnetic and inductive coupling, effective only at short distances. Upon touching the KIT, data is transmitted to the mobile phone. In the closed-loop system, this data is then securely sent from the mobile phone to a designated website, allowing for remote monitoring of the patient's blood pressure. The work in [52] proposed a method to monitor a patient's ECG waves from any location globally using the IOIO-OTG Microcontroller. An Android application was developed specifically for ECG monitoring, where the IOIO-OTG microcontroller connects to the Android phone via USB cable or Bluetooth. After data collection, the ECG waves are transmitted to the Android app, which allows for both monitoring and storage of the ECG data. The work in [53] concentrated on monitoring body temperature using a Raspberry Pi board within a cloud-based system. Their approach involves using the Raspberry Pi to monitor body temperature, with the data being transmitted through wireless sensor networks (WSN) to a cloud-based platform, where it can be accessed for temperature monitoring. The work in [54] introduced a system for monitoring body temperature using an LM35 temperature sensor connected to an Arduino Uno board. A website in SQL database format was created and linked to the Arduino Uno, enabling sensor output to be sent directly to the website. Through this platform, users can log in and monitor body temperature remotely. The work in [55] explored the monitoring of ECG, respiration rate, heart rate, and body temperature using a system where various sensors are connected to a PIC16F887A microcontroller. After gathering data from the sensors, the information is uploaded manually. To facilitate monitoring, an application, and a webpage were developed for health status tracking. The work in [56] discussed a system that monitors temperature, blood pressure, and heart rate of patients. In this setup, sensors are connected to a microcontroller, which is also linked to a GSM module. Once data is collected, an SMS is sent to the doctor if any values are concerning. The work in [57] described a method for monitoring ECG waves using an AT Mega 16L microcontroller. The system employs a Zigbee module to transfer ECG data to the nearest connected system for further analysis. The work in [58] developed a system for controlling and monitoring home appliances via an Androidbased smartphone. An Arduino Uno board is connected to home appliances like lights and fans. An Android application was created for this smart home system, enabling users

to control and monitor appliances remotely from anywhere in the world using the internet. The work in [59] focused on monitoring body temperature and heart rate using a C8051F020 microcontroller. Wearable sensors collect data, which is then transmitted to the microcontroller. A Zigbee module connected to the microcontroller transmits the data to the nearest receiver.

Typically, PMS architectures consist of either three or four layers, depending on the system design. These layers share common concepts across different PMS implementations. The PMS architecture refers to the organizational structure that defines how the system operates and how different IoT components interact with each other. This architecture typically consists of multiple layers, each serving specific functions and contributing to the overall operation of the healthcare system. PMS architecture can be categorized into different layers which are physical, logical, and application layers. The physical layer includes a number of different sensor nodes that are operating within a wireless network. Then, the logical layer comes to processes the data collected from the sensors in the physical layer. Finally, the logic layer deals with media access control and mind-to-mind communication. The application layer is responsible for determining the decisions based on the data processed in the logic layer. The work in [44] defined the PMS by using a three-layer architecture, which involves BSN, widearea-network (WAN) and personal-sensor-networks (PANs). Physiological parameter sensing is the responsibility of BSN, while the final layer addresses the interrelated medical services. PSNs, or personal surveillance networks, are made to record and process contextual information about people and their environment. Through personal service applications that function as a gateway layer, PSNs are important in facilitating integration between the first and third layers. The work in [60] divided the architecture of the residential environment in an e-healthcare system into four layers. In this architecture, layer 1 refers to the BAN layer, which includes different types of sensor nodes operating within wireless networks. Layer 2 comes to have user interaction devices. This layer works as an access-point layer to represent the user interaction interface. The medical information is gathered from layer 2 and transferred to layer 3 to prepare for the final destination using home networking possibilities. The role of Layer 3 is to filter and analyze the data collected, which is automatically connected to the Internet. Finally, Layer 4 provides the required healthcare to the patient based on delivered analyzed data. Another approach PMS architectures is proposed in [61]. In this PMS architecture, four layers are proposed which are: sensors, data acquisition, transmission, and database layer. In the proposed architecture, the Wi-Fi communication technique was used to connect the sensors to the smartphones, while 5G technology was used to connect smartphones to a cellular network, which in turn sent the data to the database. Besides, machine learning (ML) algorithms are proposed to classify the data for diagnostic purposes. The first layer was used to connect sensors and interface them wirelessly to layer 2. Layer 2 is represented by the patient's smartphone. Using a mobile application, the collected data is sent to the base station via the 5G network through layer 3, which is the transmission layer. The final layer, which is the database layer 4 was utilized as a processing unit that stores, processes, and classifies the data using machine learning (ML) algorithms. Furthermore, several works have investigated different PMS architectures, see. e.g., [40], [62], [63], [64], [65], [66], [67].

In this paper, we propose to use five 5 layers PMS architecture. FIGURE 5 illustrates the proposed general architecture for remote PMS, which is composed of a sensing layer, wireless communication layer, edge/fog computing layer, cloud computing layer, and application/action layer that involves a user interface and medical server layer. As shown in FIGURE 5, the data is obtained by different types of sensors and wearable devices, then it is transmitted using a short-range communication protocol to the nearest gateway, such as a display, a smartphone, or a computer. After that, the processed signal is transmitted to a remote server that is used by the healthcare specialized person, which is usually placed in a health institution. End users are the one benefit from the PMS, which could be patients, medical staff, hospitals, doctors, government organizations, clinical research institutes, and manufacturing companies. The general architecture of PMS can be summarised as follows:

- Things layer (sensing layer): This layer is the lowest level of the architecture where data is captured from sensors and devices attached to the patient. Specifically, this layer involves collecting of the patient's vital signs by using interoperable wearable medical devices, such as wearable sensors, EMG, ECG, SpO2, heart rate monitors, temperature sensors, ambient sensors, blood pressure sensors, and many other sensors and devices. The sensing layer collects real-time health-related data from patients and transmits it wirelessly to the higher layer.
- Communication/Networking layer: This layer acts as a gateway layer where the data collected from sensors and wearable devices is processed and aggregated locally before being forwarded to a higher-level layer or the cloud. This layer can manage device connectivity. It might also facilitate the transmission of processed data to higher-level systems or healthcare professionals. This layer involves networking protocols to ensure that healthcare data is transmitted securely and efficiently. In particular, the communication and networking layer often communicates with devices using short-range wireless protocols like Bluetooth, Zigbee, and Wi-Fi. Then, the data is forwarded to the cloud using Wi-Fi or cellular networks.
- Edge/Fog computing: This layer serves as an intermediate hub that can pre-process the data, and perform basic analytics. Edge computing can be used to enable programmable decentralized networks, and



FIGURE 5. The general multi-layer architecture of PMS in healthcare systems.

hence, facilitating effective processing and management of data at the network's edge rather than at the centralized cloud. This is particularly useful for delay-sensitive applications such as PMS. Noting that fog computing is an edge computing technology that brings cloud computing closer to the locations where data is generated and collected by extending its capabilities to the network's edge. Such distributed computing technique enables the processing and analysis of data to be performed closer to the data source, hence reducing latency, conserving bandwidth, and improving overall system efficiency.

- Cloud computing layer: IoT utilizes a distributed platform to process and store data. Data can be obtained from the communication/gateway layer. Alternatively, data from sensors and wearable devices can be sent directly to the cloud layer for processing, storage, and analysis. Cloud-based platforms provide scalability, flexibility, and accessibility, allowing for centralized data management and advanced analytics. The devices typically communicate with the cloud using WiFi, cellular networks, or other long-range wireless technologies. In the cloud layer, the raw data is processed and analyzed. Algorithms may be applied to detect patterns, anomalies, or specific health parameters.
- Application/action layer (medical server): The processed data is presented to healthcare providers or integrated with electronic health records (EHR) systems. This layer enables visualization, alerts, and decision support tools for clinicians. This layer may involve self-management, active assisted living (AAL), activities behaviors, emergency detection, disease prevention, and mobile health (m-Health) [68] or electronic health (e-Health) [69].

A complete review on the significant components used in PMS architecture is provided in the following section. This would reflect each point in architecture provided previously.

IV. SENSORS AND WEARABLE DEVICES IN HEALTHCARE SYSTEM

This section discusses the application of sensors, which is considered the most essential part in PMS healthcare architecture. Then, a description of the wearable devices and their use in PMS.

A. SENSORS FOR PMS

Typically, the first layer in PMS healthcare architecture is represented by sensor nodes. Sensors are made to be implanted beneath the skin (in-body), sewn into clothing (wearable), or applied to the human body as tiny patches (onbody) [60]. In general, sensor is a device that receives signals and responds to a stimulus. An electrical value is translated from a nonelectrical value via a sensor [70]. A sensor is defined as a computing device that can be installed on people, animals, and objects. Sensors are ubiquitous, with applications spanning homes, offices, shopping centers, and automobiles, embodying a fundamental component of the modern world. Sensors can sense, transmit, process information, and perform different works such as motion detection, image sensing, voice control, environment perception, physiological signal monitoring, and gesture recognition.

In PMS, various sources are used for data acquisition and gather information related to the patient, environment, activities, and behaviors. Sensors are the most important part in the PMS, which are used to collect information on the patient's condition. They are smaller in size, faster in process and gathering information, and cost-efficient [66]. It is important to mention that sensors are considered as the core element or fundamental component of any wearable device [71]. Hence, it is essential to delve into the specifics of sensors and explore their fundamental types.

In wearable technology, sensors play a pivotal role in capturing various physiological and environmental data that are essential for monitoring health parameters or activity levels. Different types of sensors, such as accelerometers, heart rate sensors, temperature sensors, and ECG sensors, are integrated into wearable devices to collect specific data points. Understanding the characteristics and capabilities of these sensors is essential for optimizing the accuracy and effectiveness of wearable devices in healthcare applications. Studying sensor technology within wearables not only enhances our comprehension of device functionality but also guides decisions on device choice, data analysis, and the creation of modern healthcare innovations. Hence, this review paper provides an explanation of the different types of sensors to provide researchers with the important information they need.

The emergence of the first thermostat in 1883 is often regarded as the inception of modern sensor technology. Since then, sensors have evolved significantly, adopting various principles and forms. Early sensors were relatively straightforward, measuring specific quantities and generating mechanical, electrical, or optical signals in response. However, over the past decade, the landscape of sensor technology has been revolutionized by advancements such as computing power, pervasive communication networks, internet connectivity, mobile smart devices, and integration with cloud computing. These developments have greatly enhanced the capabilities of sensors, enabling sophisticated applications in healthcare, wellness monitoring, and environmental sensing [72]. The evolution of sensor technologies into pervasive sensing environments poses fascinating not merely through integration them into sensing networks but through the way of adapted these technologies to operate in diverse collective sensing applications. Network technologies allow sensors to be connected and integrated into home infrastructure. Every sensor is performing one or more tasks concurrently [73].

Sensors have several forms of properties of the input signals and electrical output signals. They can detect any small change in the sensed quantity of the input and produce a change in the electrical output using their measuring capabilities. It is important to distinguish between sensors and transducers. The transducer converts any type of energy into another, while the sensor converts any type of energy into electrical energy. For example, a loudspeaker is a transducer that receives the electrical signal and converts it into acoustic waves. Sometimes, a Transducer can be used as an actuator in different applications, which converts electrical signals into nonelectrical energy (the opposite of a sensor). An electric motor that transforms electrical energy into mechanical action is an example of an actuator [74].

Sensors can be classified into different types and can be made based on different aspects such as uses, applications, and material, and based on their characteristics such as accuracy, range, and cost. In general, the data generated by sensors has also different formats. This format is classified as numerical, categorical, graphics, and video. Therefore, PMS can be categorised into two partitions. One partition called vision-based approach and the other called sensor-based approach, which is related to the formats and sensor types as discussed in [73]. Different types of sensors with their applications are presented in [72]. Besides, Table 1 presents a comparison of the most common types of sensors [75], [76], [77]. Noting that sensors such as sound, light, smoke, color, and seismic sensors are typically employed in various innovative ways to capture physiological data by monitoring changes in the environment that correlate with specific bodily functions. For example, sound sensors can be used to detect breathing patterns or heartbeats by capturing the acoustic signals generated by these activities [78]. On the other hand, light sensors can measure changes in skin tone or detect blood flow by monitoring variations in light absorption, often used in pulse oximetry [79], [80], [81]. Smoke sensors, though more commonly associated with detecting fire, can be adapted to monitor respiratory functions by sensing changes in air quality or exhaled gases, indicative of metabolic processes [82]. Color sensors are instrumental in analyzing changes in skin color, which can reveal information about oxygenation levels or emotional states. Seismic sensors, traditionally used for detecting vibrations, can be finely tuned to monitor subtle movements such as those caused by heartbeats or muscle contractions [83]. By leveraging the sensitivity of these sensors, physiological data can be gathered with high precision, allowing for continuous and non-invasive monitoring of vital signs.

Sensors can be divided into passive and active. A method of using passive and active sensors in road extraction is presented in [84]. The passive-type sensor does not require an additional energy source. It is able to generate an

TABLE 1. Different types of sensors.

Sensors	Action	Advantages	Disadvantages	
Tempe-	Detect thermal parameters	• No need to use a reference tempera-	Error of self-heating from applied	
rature	and record signal temperature	ture.	power and challenging to adjust.	
sensor	changes.	• Large response time and easy display.		
Position	Detect the position of an object.	• Accurate and predictable measure-	The target object's metal kind deter-	
sensor		ment.	mines the range.	
		• Higher switching rate and suscepti-		
		bility to noise.		
Sound	Detect the intensity of the sound	Real-time sound manipulation is sim-	Sound files require more memory	
sensor	and sound pressure waves.	ple and doesn't require cabling.	size and have limited coverage area.	
Light	Transforms detected photons of	Requires lower power, available in dif-	Nonlinear characteristics, temper-	
sensor	light energy into electrons for	ferent shapes and Sizes., quick re-	ature sensitive, and vulnerable to	
	use in devices like photodiodes	sponse time, and low cost.	surges and spikes.	
A 1	and resistors.			
Accelero-	Detects the acceleration of an	Good response at higher frequencies,	Sensitive to high frequency and re-	
meter	of gravity and calculates the ab	amell size	quires external power.	
	of gravity, and calculates the ob-	sman size.		
Infrared	An infrared (IR) sensor con	Operates with low power strong poise	Requires line of sight Deployment	
sensor	sists of two packs. Transmitters	immunity and detects presence or ab-	and has I imited range Affected by	
5011501	transmit the rays in the infrared	sence of light	environmental conditions	
	spectrum and the receiver re-			
	ceives the IR spectrum range.			
Pressure	Assign signals to the inputs	Low cost and high output signal level.	High hysteresis, sensitive to vibra-	
sensor	of control and display devices		tions and movable contacts.	
	based on the measurement of			
	gas or liquid pressure.			
Ultrasonic	Sensing and measuring the dis-	Greater sensing distance, resistance to	Sensitive to variation in the Tem-	
sensor	tance of a particular object.	dust, rain, snow, and other environmen-	perature, cannot work in a vacuum	
		tal factors, and the ability to sense all	and sensing accuracy is affected by	
		elements in dark situations are its main	soft materials.	
		features.		
Smoke	Utilised to detect the existence	• Simple and low-cost technology.	Requires air or oxygen to work, nar-	
and gas	and characteristics of various	• Wide measurement range.	row or limited temperature range.	
	gases and transmit signals in the	• Increased dependability, sensitivity,		
	form of output signals to the	and resolution.		
Humidity	Determine the amount of water	Does not require much maintenance	• Sensitive to dewing and Sub	
sensor	present in the surroundings and	Flexibility to use	stances	
501501	translate these findings into sig-	No aging effects	limited accuracy and measurement	
	nals suitable for input stimula-		range.	
	tion.			
Color	Uses a receiver to pick up light	Simple to implement and easy to ad-	Sensing range affected by color and	
sensors	reflected off the detecting item	just configurations without even repro-	reflectivity of target.	
	after light is emitted from a	gramming the sensor device.		
	transmitter.			
Chemical	Transmits chemical information	Linear output, low power requirements	Narrow or limited temperature	
sensors	from a chemical reaction.	and good resolution with excellent re-	with excellent re- Range with short or limited life	
		peatability and accuracy.		
Seismic	Records and amplifies minute	Detects lateral and vertical variations in	Data processing is time-consuming	
sensor	motions of the earth in addition	velocity and produces detailed images	and equipment is expensive.	
	to measuring them.	of the Subsurface.		
1			Continued on next page	

TABLE 1. (Continued.) Different types of sensors.

Sensors	Action	Advantages	Disadvantages
Touch	Function as switches, and when	Touch interactions are natural and fa-	Touchscreens are susceptible to ac-
sensor	the sensor's surface is con-	miliar to users accustomed to smart-	cidental touch and more susceptible
	tacted, the circuit's current be-	phones and touchscreens and elimi-	to scratches and cracks.
	gins to flow, just like it would in	nates the need for protruding buttons.	
	a closed circuit.		
Magnetic	A response by generating a pro-	Contactless Operation with high-Speed	Limited to conductive materials and
sensor	portionate output to the pres-	Response and high Resolution.	higher cost.
	ence or absence of a magnetic		
	field, including flux, strength,		
Ontical	and direction.	Descenshly priced production viewel	Due biocompetibility in fluores
optical	Quantines biological of chemi-	estimation simplicity in design speedy	cance and colorimetric sensors. Pe
5015015	tions in colour scattering fluo-	optimisation, and adaptability elevated	quires bulky UV lamps or flu-
	rescence and light absorbance	sensitivity corrosion-resistant and suit-	orescence spectrophotometers and
	rescence, and right assorbance.	able for use in challenging conditions.	bulky size.
Proximity	Used to find adjacent items	It can detect both metallic and non-	They are affected by temperature
sensors	without coming into direct con-	metallic targets, It has good stability	and humidity and difficulties in
	tact with them.	with high speed provided. Low cost and	designing with limited operating
		power consumption and useful and can	range and expensive than inductive
		help with many security problems	and capacitive sensors.
Force	Convert applied mechanical	Accurately measure the magnitude and	High-precision force sensors can be
sensors	forces such as compressive	direction of applied force and provide	expensive and certain environmen-
	and tensile forces-into	real-time data on force variations, en-	tal factors like temperature fluctu-
	digital signals whose values	abling adjustments in processes for op-	ations might affect sensor perfor-
Flow	Used to measure a fluid's flow	Accurate and reliable data on the rate	The cost of flow sensors can very
sensors	such as a liquid or gas and send	of fluid movement through a pipe or	depending on the type accuracy
50115015	the controller output signals.	channe and come in various types suit-	and features. High-precision or spe-
	···· · · ···· · · ··· · · ··· · · · ·	able for measuring different fluids (liq-	cialized sensors can be expensive
		uids and gases), flow rates (low to	and may require periodic mainte-
		high), and pipe sizes and can be inte-	nance or calibration to maintain ac-
		grated with data acquisition systems.	curacy, especially in harsh environ-
			ments or with continuous use.
Flaw	Used to display irregularities on	• Allow for early detection of defects	• Analyzing sensor data and
sensors	surfaces or different underlying	before they become critical failures,	interpreting results often requires
	materials in a variety of manu-	enabling preventative maintenance and	trained personnel with expertise.
	facturing processes.	avoiding costly downtime or accidents.	• Depending on the sensor type,
		• Flaw sensors are used in various	deep flows can be detected within a
		motive construction and nower gener-	material
		ation	material
Electro-	Transform the redox reaction's	Low waste fabrication and possibility	Requires external power sources
chemical	impact on electrode surfaces	of large-scale production.	and requires amplification tech-
sensor	into electrically readable signals		niques for highly sensitive signals.
	that indicate variations in con-		
	ductivity, current, and potential.		
Piezo-	Investigated for energy-	Broad sensing range, high-frequency	Charge leakage and the only possi-
electric	harvesting solutions and	response, and simple construction.	bility of dynamic sensitivity.
sensor	self-powered sensors in		
	biointegrated devices.		

electric signal directly in response to an external stimulus. Examples of passive sensors are thermal, infrared, electric,

and chemical. On the other hand, an active sensor type needs an external energy source for their response, (known

as an excitation signal). Examples of active sensors are thermistor sensors and resistive strain gauge sensors. The work in [85] provided an extensive review of sensor technologies and their transformative impact on our lives across various fields. The author discussed how sensors are able for detecting changes in the environment and collecting signals, hence enabling a wide range of applications spanning lifestyle, healthcare, fitness, and manufacturing. Specifically, the paper highlighted sensor applications in healthcare, where drug-delivering sensors can be used to aid medication adherence by reminding individuals to take medicine and providing doses at specific times, benefiting older adults, athletes, and at-risk patients. Additionally, the review outlined key industrial trends such as ultrasound, radar, and optoelectronic solutions driving sensor innovation, emphasizing their critical role in modern business operations and everyday activities. The work in [86] explored the diverse and rapidly evolving field of sensors driven by high demand and continuous technological advancements. The authors discussed the electrochemical sensors that is known for their affordability and versatility in detecting various analyses and are extensively employed across industries like agriculture, food, oil, environment, and healthcare. The review highlighted the appeal of electrochemical sensing due to its versatile reporting signals (voltage, current, power output, impedance) and low detection limits attributed to Faradaic and non-Faradaic currents. Additionally, it discusses recent advancements and applications of electrochemical sensors, particularly emphasizing the role of nanomaterials. The work in [87] presented a comprehensive review of advancements in soft sensor design and implementation, focusing on their critical roles in industrial process monitoring, control, and optimization. The author also discussed how new theories, techniques, and information infrastructure have improved soft sensor performance while also how to address some related technical challenges. Furthermore, the author discussed the most recent developments in the soft sensor design, offering insights from a systems and control perspective, to provide up-to-date information related to this research. The work in [88] reviewed the integration of artificial intelligence with health monitoring sensors to improve healthcare capabilities, addressing some technical challenges such as noise, data processing, and feedback control. Besides, this paper explored advances in wearable and implantable sensors for monitoring vital signs, soft electronics for therapy, and volatile organic compound detection. Furthermore, this paper discussed some recent developments in artificial intelligence-enhanced humanmachine interfaces and self-sustainable sensor systems, highlighting the potential for more intelligent and secure healthcare services in future biomedical applications.

There are three classes of interconnected networks namely as PSN, BSN, and multimedia devices. Such networks can be deployed within living environments or integrated into various household objects such as sofas, tables, beds, chairs, or floors, all equipped with pressure sensors. By monitoring subjects within their environment and interactions with these objects, PSNs offer valuable insights into the performance levels of daily living activities. The data provided by PSNs enables a deeper understanding of individual behavior and facilitates the development of personalized solutions for healthcare and lifestyle management [73]. Usually, the first layer of PMS architecture is termed BSN. A BSN can be defined as a wireless network obtained by deploying different sensors in and around the human body [89], hence it is a network of wearable sensors that are used for monitoring patients [90]. Miniaturised wearable or implantable wireless sensors have been employed in the introduction of the BSN concept. Imperial College London proposes the BSN node, a BSN hardware development platform, to support BSN research and development. BSN nodes offer a flexible development environment for ubiquitous healthcare applications because of their low power consumption, compact size, and flexible design. The sink node in a BSN gathers data and sends it across the Internet for sharing. Social welfare, emergency treatment systems, diagnosis services, and patient direct care can all benefit from having a BSN [91]. Basically, the identification of the sensor node is defined by a unique ID for each device. For example, in a single BAN, two motion sensors are used: one for tracking hand motion and the other for tracking foot movement [92]. BSNs are moving towards intelligence and multi-technology integration. Despite the existing challenges associated with BSNs, their future holds significant promise as they have the potential to deeply transform human-machine interactions.

The widespread adoption of home appliances integrated with sensors is paving the way for interactive healthcare environments. These integrated appliances are electronic or electrical gadgets that have several uses in the house. Devices such as TV sets, speakers, phones, cameras, and microphones create a platform for data sharing between people and healthcare systems. These devices enhance user engagement with health applications, serving as new sources of contextual data and platforms for guidance and counseling. Multimedia devices, such as cameras and microphones, are central to these approaches, enabling visual and audio sensing for monitoring daily activities [73]. Wireless sensors, including smartphones, smart cameras, and devices like Raspberry Pi with cameras, are instrumental in monitoring chronic disease patients, comatose patients, and newborns for smart caregiving and independent living in smart homes. These sensors facilitate measurements such as respiratory and cardiac rates using air quality sensors, integrated toilet seats with ECG monitors, cardiac and respiratory monitors, and smart beds equipped with sensor-enabled pillows and magnetic switches. Vision-based methods extend further to include posture recognition, human presence detection, movement and fall monitoring, and tracking complex activities. While multimedia-based approaches offer rich contextual insights, they are challenged by computational demands and privacy concerns. Data acquisition methods vary across sensors, and the heterogeneous sensor set provides primarily raw,

low-level data that is imperfect, uncertain, and of limited meaning. Thus, further advancements are necessary to develop higher-level approaches for delivering comprehensive healthcare services. Besides, advanced signal processing techniques such as those presented in [93], [94], [95], [96], and [97] can be used to obtain efficient healthcare systems.

When it comes to creating and retrieving raw sensor data-that is, information about the patient and their surroundings-physical sensors are thought to be the most often employed type of sensors [73]. The common types of sensors used for the purpose of health sign activity monitoring, which are generally attached to human body, are provided as follows [98], [99]. Electrocardiogram (ECG) sensor is used for heart rate monitor. Electromyography (EMG) sensor is used to track muscle contraction. EEG (electroencephalography) sensor is used for brain electrical activity monitoring. Glucose sensor is used for measuring and monitoring glucose levels in the body. Pulse oximeter (SPO2) is used to measure oxygen saturation of blood. Body temperature is used to track temperature sensor. Respiration sensor is used for detecting and measuring respiratory parameters such as breathing rate, volume, or patterns to monitor respiratory function and provide insights into respiratory health. Finally, motion sensors (accelerometers) used to estimate user's activity.

B. WEARABLE SENSOR DEVICES

Data acquisition plays a crucial role in PMS by utilizing smart health devices, primarily wireless sensors and wearable devices, which gather comprehensive patient data. Wearable devices like digital watches, smart clothing, and fitness tracker bands collect a diverse range of health information, enabling continuous PMS and data collection for wearable health monitoring systems) and general health monitoring systems. Specialized devices like smart vests are used for non-invasive physiological monitoring, including ECG, galvanic skin response, body temperature, and blood pressure assessment. Such data acquisition methods not only enhance athlete performance and support disabled individuals but also enable personalized healthcare interventions based on real-time patient data.

Since Steve Mann, who is recognised as the father of wearable computing, unveiled the first Linux-based wristwatch at the 2000 IEEE International Solid-State Circuits Conference, the number of wearables has increased dramatically over the past ten years [100]. The first wearable sensor was called Holter monitoring, which was developed in the late 1940s and used in clinical settings in the 1960s. It is a portable gadget that tracks the central nervous system's electrical activity continually. The field of wearable sensors has advanced during the last ten years, having begun to take shape in the first part of the 20th century. Wearable devices have been introduced as key enabling of IoT-based healthcare systems, which attracted much attention due to their various functionalities. Besides, the wearable devices become available in the market at an acceptable price [101]. Wearable devices have many advantages and are becoming more and more popular because they make life easier. Wearable devices have been used in different applications such as environmental detection, physiological signal monitoring, human activity recognition, and most importantly healthcare monitoring. An example of wearable devices are smartwatches, smart glasses, and fitness trackers. Particularly, these devices integrate low-power sensors to detect movement and other physiological signals. As a result, research and commercialization are now focused on wearable sensors as functional parts of wearable technologies [71]. With the recent advancements in IoT and artificial intelligent technologies, wearable sensors have been developed and employed widely in the scientific and industrial communities and receiving significant attention in sophisticated scenarios such as navigation systems, biomedical applications, Google Glass, consumer goods and smart clothing. The combination of IoT, augmented reality, and wearables can create new paradigms that may potentially change the way people experience the world [102]. Proactive personal health management involves monitoring patients outside of the hospital setting, often using wearable sensor devices. This approach allows individuals to actively track and manage their health status remotely.

The wearable devices can be placed on-body biosensors, which should unobtrusively measure significant physiological signals like blood pressure, body movement, heart rate, skin, and body temperature [60]. Wearables can be worn on the body itself, concealed in clothing, or housed in semi-rigid objects like headgear, gloves, insoles, and smart watches. In PMS, wearable devices cover a wide and important range of applications. The working and characteristics of wearable devices in healthcare fields are mainly based on the observation of physiological data from the user's body [71]. These wearable devices in PMS are considered simple in use, low-cost, independent devices targeting personal use without the need for medical professionals. These properties make wearables devices widely deployed, and in turn, has led research attention at a much faster pace to this field [103].

Typically, wearable devices can be divided into three groups according to the position of wearing: hand-worn, limb-worn, and head-worn. Different considerations and requirements of such devices should be taken into account such as the cost, portability, usability, wearability, intelligence, and performance. TThe development of wearable sensors grew quickly as a result of the growing need for telehealth and real-time health monitoring, with a compound annual growth rate of 18.3% from 327.6 million in 2021 to 1,487 million in 2030 [29]. The statistics predict that by 2025, there will be multibillion wearable sensors, where over 30% of them are considered as new types of sensors that are just beginning to emerge [104]. There is a greater exchange of data across medical equipment when wearables are used. The ecosystem addresses these issues in wearables interoperability & intelligence by including different points. Standards developers, payers, providers, healthcare delivery



FIGURE 6. Key features of wearable devices in PMS.

organisations, network service providers, and manufacturers of IT and medical device hardware, firmware, software, and services [105]. Nevertheless, there are a number of issues with wearables and medical IoT connectivity and intelligence, including the possibility of wearable and implantable hacking.

In general, the two key challenges that face wearables devices are security issues and power efficiency [40]. Practically, wearable monitoring equipment is very efficient for patients enabling him/her to monitor his/her own health status anywhere. Wearable medical technology is expected to keep expanding, particularly in rural areas. Additionally, the vast economic market helps to fuel the quick growth of these gadgets [106]. Therefore, the commercially available wearable devices and prototypes for medical IoT need to be classified to distinguish the best devices for each patient status that is monitored [103]. It is worth mentioning that wearable sensors with other diagnostic platforms such as point-of-care devices and lab-on-a-chip systems are integrated with biosensors. In order to monitor the concentration of different analytes in biological fluids including blood, saliva, and urine, biosensors are widely employed in medical diagnostics. Wearable biosensors, or devices with built-in biosensors, can save healthcare expenditures [107]. Additionally, wearable biosensors are a useful tool for identifying bacteria, spotting dangerous compounds in the environment, and diagnosing ailments in people. The following are some of the key features of wearable devices [108], [109], [110]. FIGURE 6 summarizes the key features of wearable devices in PMS.

• Wearability: This characteristic is mostly dependent on the size of the sensors, the power supply specifications, and the reduction of on-body hardware. Future developments could address the used hardware by incorporating the parts into e-textiles. Textiles' special qualities of being lightweight, pliable, and comfortable to wear make them ideal for wearable technology.

- Low cost for various applications: The cost-effectiveness considering the initial investment, training, and maintenance will surely influence the integration of the wearable devices into PMS.
- Usability: The functionality of these devices such as data collection, transient information collection, and user-friendly interface are required to satisfy the specific clinical need and measure the important signs of the patient.
- Power consumption and ease of fabrication: These are major challenges in the design of wearable devices as long-term maintenance-free operation is required. Hence, advanced methods for low-power processors are essential.
- Performance and reliability: Verification in real-life scenarios provides essential information based on its usefulness and reliability. Wearable devices should demonstrate specific tolerances for patient activity in the environment to contribute the reliability through the identification of mistakes, avoiding misleading the collected data.

Wearable device innovation is being propelled by advancements in wireless communication, energy harvesting, and sensor networks, in addition to the previously mentioned factors. In [103], the commercial wearable existing products were classified under three categories: 1) Accessories, 2) E-Textiles, and 3) E-Patches. A summary of commercial products and research prototypes are surveyed.

Medical sensors operating in PMS might have varying principles depending on factors like photoelectric, hall, and piezoelectric effects. The sensors attain the physiological parameters of the patients and normally consist of sensitive and conversion elements [106]. In the design of wearable devices, the significant issue is based on the sensors that are embedded within the wearable devices. Different types of sensors are used for this purpose such as pulse oximeter, EEG, breathing sensor, heart rate monitor or ECG, respiration rate, movement sensor, EDA, SpO2, and blood pressure [111], [112]. Besides, several historical examples of wearable sensors are provided in [113]. Traditional sensors measured the heart rate alone, where this sign is considered the most important and almost all medical sensor nodes support it, then the photoplethysmography appears to measure the SpO2 besides the heart rate [114]. Chemical and biological materials are integrated with optical systems, microfluidics, electronics, and micro-machines to create wearable technology-comfortable, wireless, battery-operated systems that can provide data on a regular basis [115], [116]. The human body can serve as a transmission channel for wearable devices, or they can connect with one another via a particular transmission medium like Wi-Fi, BLE, Zigbee, and other previously covered communication protocols. They

are able to gather, sort, and store the subject's long-term physiological and activity data [117]. In essence, there are two main strategies for implementing wearable technology: using wireless technology and electronic textile solutions, as covered in [92]. Several research works have investigated wearable devices and their applications highlighting their advanced technologies, significance, industrial communities, materials, target analytes, and design issues, see, e.g., [114], [117], [118], [119], [120], [121]. FIGURE 7 presents the most common types of wearable devices.

V. DATA TRANSMISSION AND NETWORKING

A paradigm change has occurred in recent years that makes it possible to virtualize and softwarize networks, or create programmable decentralised networks [122]. By segmenting the network into several communication layers, this can be accomplished. These layers can help mitigate the delay sensitivity of IoT applications by bridging the lengthy propagation distance between the end-user and the cloud centre. As was already indicated, patient data and their physical activity can also be recorded using wearable technology and fitness watches. At the beginning, wearables and sensors are used to collect patient health data, and connected devices exchange data with other devices while transmitting healthcare information. IoT devices that are worn inside the body gather medical data, including blood pressure, glucose levels, pulse rate, ECG, and cholesterol.

It is occasionally possible to perform data preparation, which entails removing noise and missing values. Thus, removing noise from data cleaning results in a higher diagnosis process detection rate. The median studentized residual approach, which thoroughly analyses the relationships within the dataset, can be used to filter out undesirable data. This method leads to improved detection rates in the diagnostic process. Initially, missing data is addressed by replacing them with median values across rows and columns. Subsequently, data normalization is performed to scale values between 0 and 1, reducing complexity in the diagnostic process. This normalization involves adjusting data using various distribution techniques.

It is necessary to send the data that sensors capture from the patient to the closest data collection or processing node. Wireless Body Area Networks (WBANs) are one technology that is helpful for moving data from the sensors to the nearest processing node [123]. WBSNS will be discussed in the next subsection.

A. WIRELESS BODY AREA NETWORKS (WBANs)

Various components work together in the IoT architecture to enable different solutions for the end user. An IoT-enabled network of linked devices that perceive critical data in real-time makes up the PMS-based healthcare application. In particular, PMS makes it possible to regulate end-user applications through analytics, device connectivity, data transfers, and real-time data collecting. WBANs are among the technologies that enable data flow from sensors and drive communication to the closest processing node in an IoT architecture [124]. The most prominent application area of WBAN is PMS [124]. Because WBANs measure physiological activity in humans, which varies more periodically, the application data streams show rather constant rates [39]. The tiny, samrt wireless sensors that make up BANs are in charge of collecting and relaying to carers the vital indicators of a patient [125]. WBAN can be used in transfer the information of the patients into a wide communication ranges. It should be noted that the terms WBAN, WBASN, and WBSN are used interchangeably in the research based on the work context. AThese terms mean the same type of networks which consist of sensor nodes to check and monitor the human physiological conditions of patients through their remotely accessed signals [126]. IEEE standards like IEEE 802.15.6 and IEEE 802.15.4j, which are especially designed for medical WBANs, are used by WBAN. WBANs provide dependable communication, speed, accuracy, and energy efficiency between sensors and actuators placed inside, on top of, or next to the human body. By leveraging different sensors to monitor patient health vitals, WBANs can contribute to reducing healthcare costs and enhancing the quality of care [123]. Broadband access networks (WBANs) facilitate a range of data rates, from ultra-wideband (15.6 Mbps) to narrowband communication (75.9 Kbps). Wireless Local Area Networks (WLANs), Zigbee, Bluetooth, mobile networks, Wireless Personal Area Networks (WPANs), and Wireless Sensor Networks (WSNs) are among the other wireless technologies with which they are compatible. Sensors that are dispersed geographically are typically used in WSNs to observe, record, and monitor environmental and physical variables. They require an infrastructure made up of relatively small nodes, which are the core components of WSN and are grouped in groups ranging from a few tens to hundreds or thousands of nodes. They are used to collectively transfer observed data via the network to a central location. where one or more sensing devices are installed on each node. The network architecture is the second part of WSN. A vast region is covered by a large number of sensor nodes that are then networked together. There is communication between sensor nodes and with a base station [127]. The development of WSN was first motivated by military purposes such as battlefield surveillance. WSNs are used in different fields like industry and medical applications [127]. Note that WSNs and Wireless Body Area Networks (WBANs) share similarities in wireless communication technology but differ in their specific applications. WSNs are designed for broader environmental monitoring and data collection, while WBANs focus on wearable sensors for health monitoring and medical applications, often integrated into clothing or implanted on the body. In healthcare, WBANs play a crucial role in improving patients' quality of life by enabling continuous monitoring and personalized healthcare interventions. Three categories can be used to describe WBAN: beyond-WBAN communication, intra-WBAN communication, and

IEEE Access

FIGURE 7. Illustration of the the most common types of wearable devices (reproduced from [103]).

inter-WBAN communication [125]. WBANs can be either wearable, i.e., used on the body surface of a human or implantable, i.e., inserted inside the human body [128].

As stated in, a WBAN node is a standalone device with communication capabilities that falls into one of three types according to its capability, implementation, and network

role [129]. In terms of functionality, nodes include personal devices, sensors, and actuators. Personal devices gather data from sensors, process it, and can communicate with other devices or individuals through an external gateway or by activating actuators. These personal devices are also referred to as body gateways, sinks, or Body Control Units (BCUs) in certain contexts. Sensors transfer data to the personal device by capturing certain parameters from the human body either internally or outside. Actuators interact with humans based on sensor or personal device data; for example, a physician might send medication instructions via a display actuator after processing sensor data on a personal device. Nodes can be further classified based on implementation into implant nodes, body surface nodes, and external nodes. Based on their network role, nodes are categorized as coordinators, end nodes, or relays. Coordinator nodes act as gateways connecting WBANs to the Internet, and other WBANs, or serving as a central trust point. End nodes are typically sensors without communication capabilities, while relay nodes facilitate message forwarding between sensors and gateways when they are distant. WBSN mainly collects a large amount of real-time data and sends them to the cloud. The cloud, in turn sends and/or saves them in a server and then transmits them to the specialist to be analyzed based on patient statues.

B. COMMUNICATION NETWORKS AND PROTOCOLS

As previously said, PMS is a popular healthcare tool that helps physicians keep an eye on hospitalised patients, elderly individuals receiving home care, and patients with acute or chronic illnesses when they are in remote areas. Wireless network technologies control data transfer and link patient devices and sensors to distant sites. Communication methods such Wi-Fi, Bluetooth, ZigBee, cellular networks 3G/4G/5G, NFC, and satellite are used to convey the patient's signals [130]. By using efficient and reliable wireless networks, the data can be transmitted to several places like the doctors, caregiver. In order to provide data analytics and cloud-based services for biometric data gathered from physical devices and sensors, PMS models may extend big data processing to the cloud. Big data is essential because it facilitates decision-making, data analysis, and the extraction of valuable information [131]. Based on real-time data from connected devices, continuous patient monitoring offers real-time surveillance, feedback, and intervention of patient parameters. The doctor can deal with emergencies cases from a remote location by viewing the patient's report. The integration of intelligent communication technologies that can link diagnostic equipment to distant places has been the subject of some research projects. The appropriateness of every communication technology is contingent upon multiple criteria. These factors are range, data rate, and power consumption [130], [132]. Additionally, the Federal Communications Commission (FCC), a government organisation in charge of overseeing communication technologies in the US, defines sets of standards and guides that can be followed for

- Wireless local area networks, or WiFi, are thought to be the most widely utilised wireless technologies in PMS. IEEE 802.11 standards, namely 802.11b-11 Mbps, 802.11g-54 Mbps, 802.11a-54 Mbps, and 802.11n-300 Mbps, provide the foundation of WiFi architecture [133]. WiFi technology allows the devices to access the internet and communicate with each other within a local area range. WiFi technology can operate in bandwidths of 2.4 GHz, 3.5 GHz, and 5 GHz unlicensed Industry, Scientific, and Medical (ISM) frequency bands. WiFi technology is able to support short-range communications with a coverage range of up to 100m. WiFi technology also allows reliable, secure, and high-speed data communications. A network of 45 vital medical devices, including as infusion pumps, defibrillators, lung ventilators, and anaesthesia machines, have been equipped with WiFi technology, demonstrating that WiFi can be utilised safely and successfully for these devices' communication [134].
- Based on IEEE 802.15.4, ZigBee is a wireless personal area network technology. ZigBee is compatible with low data rate networks, which usually require up to 250kbps, because its data rates are far lower than those of other technologies. Long battery life of up to several years is possible using ZigBee [33]. While its power consumption is higher than Bluetooth's, a typical ZigBee transmission range can reach up to 100 metres depending on power output and ambient factors [135]. This distance can be extended up to 1,600m with ZigBee-Pro. Every Zigbee network consists of three types of devices, which are coordinator, router, and enddevice [120]. Keys for 128-bit symmetric encryption protect ZigBee networks. It features one channel in the 868MHz Europe band, ten in the 915MHz Australia and US bands, and sixteen in the 2.4GHz ISM band. Moreover, a mesh network architecture utilising ZigBee wireless technology enables higher coverage range and high-reliability communication.
- Bluetooth technology is a wireless personal area network named after King Harald Bluetooth who unified Denmark and Norway and embodies the concept of connecting diverse devices wirelessly. Originally developed by Ericsson in 1994, Bluetooth is based on the IEEE 802.15.1 standard, which is widely used for linking IoT devices, smartphones, and other mobile devices. Bluetooth operates in the 2.4-2.4835 GHz [136] unlicensed ISM band using frequency-hopping spread spectrum (FHSS) and provides a data rate of up to 721 kbps [137]. Bluetooth allows a lower transmission coverage with a coverage distance of up to 100m. Bluetooth is typically used for portable personal devices and to connect between medical devices, sensors, and

smartphones or tablets used by healthcare providers. Bluetooth-enabled devices can transmit vital signs, such as heart rate, blood pressure, and oxygen saturation, directly to monitoring systems, enabling real-time data collection and analysis. Furthermore, Bluetooth is optimised for use in loud situations, covering both the MAC layer and physical access.

- Z-Wave is a low-power and cost-effective wireless communication technology suitable for short-range applications, particularly in residential and light commercial settings [138]. With a data rate of up to 40 kbps and a coverage distance of about 30m, Z-Wave supports mesh networking, making it suitable for use in PMS. In Australia and North America, Z-Wave uses the 900MHz unlicensed radio frequency range for its operations, providing extended coverage and lower power consumption compared to higher frequency bands like 2.4GHz [139]. Z-Wave immunity to Wi-Fi interference ensures reliable communication alongside other wireless devices, offering a standardized and stable control medium for smart home automation and potentially for WBANs in healthcare applications [126]. Its ability to penetrate walls and solid objects makes Z-Wave a compelling choice for deploying sensors and monitoring devices within healthcare environments, supporting efficient and seamless patient monitoring capabilities.
- IPv6 over low-power Wireless Personal Area Networks is referred to as 6LoWPAN. This networking technology makes it possible for IPv6 packets to travel via short link-layer frames, as those outlined in IEEE 802.15.4, with efficiency. Initially, IEEE 802.15.4 low-power wireless networks operating in the 2.4 GHz range were intended to be supported by 6LoWPAN. Currently, 6LoWPAN is being deployed over low-power RF at sub-1 GHz, among other networking media. Strong AES-128 link layer security as outlined in IEEE 802.15.4 is leveraged by 6LoWPAN. Because of these features, the technology is perfect for a variety of sectors, including smart metering, residential illumination, street light monitoring and control, home automation using sensors and actuators, and general Internet of things applications involving Internet-connected devices. Depending on the benefits of IEEE 802.15.4, which include support for a vast mesh network topology, reliable communication, and extremely low power consumption, 2.4 GHz and sub-1 GHz bands are typically employed. Depending on the frequency, the 802.15.4 standard offers data speeds ranging from 20 to 250 kbps. 6LoWPAN works well over short distances, up to 100 metres. 6LoWPAN allows sensors and local devices to be connected to IP networks in healthcare systems, enabling the connectivity of several sensors [140], [141].
- EnOcean is an energy-harvesting wireless sensor technology that finds applications across diverse sectors including building automation, transportation,

environmental monitoring, and health monitoring. It operates without batteries by harnessing energy from ambient sources like light, motion, or pressure, converting them into electrical energy for use in wireless communication. EnOcean devices are designed to be ultra-low power, utilizing micro energy converters and efficient electronics to enable wireless connectivity among sensors, switches, controllers, and gateways. This technology, ratified as an international wireless standard (ISO/IEC 14543-3-10), offers significant advantages in terms of energy efficiency and maintenance-free operation. EnOcean can contribute to the PMS by enabling wireless and battery-less sensors and switches for building, home, or industrial automation applications [60]. EnOcean technology allows a long communication range of up to 300m and low-power consumption, and hence, makes this technology suitable for deploying sensors in various healthcare settings without the need for frequent battery replacements, thus reducing operational costs and simplifying maintenance tasks [142]. EnOcean devices operate at the following transmission frequencies: 902 MHz, 928.35 MHz, 868.3 MHz, and 315 MHz. EnOcean is capable of up to 125 kbps of data rate.

- The bi-directional radio frequency identification system known as radio-frequency identification (RFID) technology comprises of a tag and reader that can be interfaced with personal computers or handheld computers [143], [144]. RFID automatically recognises and tracks tags affixed to items or people using electromagnetic waves. The tag is made up of two parts: an antenna that allows the chip to use radio waves to connect with the tag reader and a chip that stores the object's unique identity. By using the tag's reflected radio waves, the tag reader creates a radio frequency field that allows objects to be identified [130]. RFID makes it easier to follow a patient's movements in real time within a medical facility. RFID adheres to the EPC (electronic product code) standard. RFID is compatible with Wi-Fi and ZigBee, among other technologies. RFID can detect objects within a range of 10 cm to 200 m and operates across a broad range of frequency bands, from 120 kHz to 10 GHz. RFID can help improve inventory control by monitoring the location and use of medical equipment and supplies. RFID technology can deliver up to 4 Mbps of data rate. Because RFID technology enables asset management and patient monitoring capabilities, it is essential to healthcare systems.
- ONE-NET is an open-source wireless networking standard designed specifically for low-power and low-cost control networks, such as those used in sensor applications, home automation, security, and monitoring. Since ONE-NET is not dependent on any particular hardware, it can be built with a variety of commercially available microcontrollers and radio transceivers from companies

including Silicon Labs, Freescale, and Texas Instruments. ONE-NET uses Wideband FSK modulation and operates in frequencies including 433 MHz, 868 MHz, 915 MHz, and 2400 MHz. It can support data rates ranging from 38.4 kb/s to 230 kb/s. With ranges of up to 100 metres indoors and 500 metres outdoors, its adaptable network topologies-peer-to-peer, star, and multi-hop-enable effective connectivity and communication coverage across indoor and outdoor situations. Because of its exceptional low power consumption, this technology can be used in battery-operated gadgets that have a five-year battery life when using AA or AAA Alkaline batteries. Furthermore, ONE-NET integrates strong security protocols, using the expanded tiny encryption algorithm to protect confidentiality and integrity of data in wireless networks [145]. ONE-NET can provide dependable and affordable wireless connectivity for sensors and devices in patient monitoring systems. It provides scalability, energy efficiency, and secure data transmission, improving the capacity for remote patient monitoring in healthcare settings.

- Long Range Wide Area Network (LoRaWAN) is a wireless communication technology well-suited for healthcare systems, particularly in patient monitoring applications. LoRaWAN enables long-range, low-power communication between sensors and gateways, making it ideal for collecting data from remote or distributed patient monitoring devices [146]. LoRaWAN technology supports secure and reliable transmission of vital signs and health data, such as heart rate, blood pressure, oxygen saturation, and body temperature, from patients to healthcare providers or central monitoring stations. LoRaWAN is considered as a robust and scalable transmission technique, which allows for the deployment of sensor nodes in various healthcare settings, including hospitals, nursing homes, and even patients' homes, ensuring continuous and real-time monitoring without the need for frequent battery replacements. By exploiting LoRaWAN technology, healthcare systems can enhance patient care by enabling proactive interventions, timely alerts for critical conditions, and overall improved patient care systems through remote monitoring and management.
- Devices can communicate with one another across a few centimetres' distance thanks to near field communication (NFC), a short-range wireless communication technique [147]. The foundation of NFC is Radio Frequency Identification (RFID) technology, which uses electromagnetic fields to wirelessly transfer data between devices when they are positioned in close proximity to one another. NFC allows for seamless and secure communication between devices in close proximity, making it suitable for healthcare scenarios where quick data exchange and authentication are crucial. Numerous uses are made possible by NFC technology, including contactless payments, access control, and

device-to-device data sharing [148]. NFC makes it simple to pair medical devices with smartphones or tablets in PMS, giving medical professionals effective access to real-time patient data.

- Medical Implant Communication Service (MICS) is a specialized wireless communication standard designed for medical implants and devices used in healthcare systems, particularly in patient monitoring applications. MICS operates in the frequency range of 402-405 MHz and enables reliable, low-power communication between medical implants and external devices such as wearable monitors or bedside receivers. MICS ensures secure and efficient communication, allowing healthcare providers to remotely monitor patients' vital signs and device status without the need for invasive procedures or physical connections. Sensor signals can be collected using the MICS band, and data can then be sent to a distant station. This process enables multiple patients to have long-range monitoring at the same time [149].
- Ultra-Wideband (UWB), also known as IEEE 802.15.3a, is a low-power, high-precision wireless communication technology that uses a broad spectrum of frequencies to send data over short distances (about 10 metres) with minimal power consumption. UWB is used in real-time applications in RF-sensitive applications, such as hospitals, and is also used for precise positioning, asset tracking, and short-range data transmission. UWB has been wildly used for healthcare systems with short-distance communication [150], [151], [152], [153]. UWB can achieve a data rate from 20 Mbps up to 1.3 Gbps. UWB is suitable for the physical layer of high data rate PANs.
- IEEE 802.15.6 is standard that is designed for WBANs, has become widely adopted across medical and non-medical wearable devices for communication. This wireless communication standard enables direct sensor-to-device communication, alleviating the need for intermediate relaying or routing devices, which enhances efficiency and reduces complexity in data acquisition processes [154]. With aid of a variety of data rates, low power consumption, and the ability to connect up to 256 wireless devices within a single WBAN, IEEE 802.15.6 effectively meets the communication requirements of wearable devices. With its ability to operate in the Narrow Band, Ultra-Wide Band, and 2.3 GHz to 2.4 GHz frequency bands, this standard supports a wide range of WBAN applications. IEEE 802.15.6 remains highly appealing for wearable and body-worn medical devices, particularly due to its suitability for short-range, low-power, and cost-effective communication services both within and around the human body. This wireless communication standard may allow a significant advancement in facilitating reliable and efficient communication for PMS.

- Sensium is an innovative low-power on-body technology designed for continuous patient health monitoring [155]. Sensium can wirelessly transmit vital sign information to smartphones, mobile phones, laptops, or PCs, allowing seamless communication with medical entities such as doctors or paramedical staff for timely intervention. Sensium sensors operate efficiently by activating only during designated time slots to transmit data, conserving energy through periods of inactivity [126]. This technology has gained widespread adoption as a leading patient health monitoring system, exemplified by its practical application in PMS during the COVID-19 pandemic. The use of Sensium patches, as depicted in the accompanying image, highlights the significance of wearable WBAN devices for remote patient monitoring, underscoring their role in facilitating effective healthcare delivery outside traditional clinical settings [156]. Sensium represents a promising solution for PMS, offering real-time data transmission and enabling proactive healthcare interventions based on continuous monitoring of vital signs.
- RuBee is a wireless protocol considered as an alternative to RFID, utilizing long wave magnetic signals instead of traditional radio frequency signals to transmit and receive data within local networks [60]. RuBee is a lowfrequency (131 kHz) device that offers notable power efficiency advantages. It can run on a single lithium button cell battery for up to fifteen years. RuBee is designed based on standard of IEEE 1902.1, emphasizes high security and robustness in harsh environments, offering advantages over RFID in terms of signal reliability and penetration through materials like steel and liquid [60]. Despite its lower data rate compared to technologies like WiFi, Bluetooth, and ZigBee, RuBee's unique characteristics make it well-suited for applications requiring long-lasting, secure, and interference-resistant wireless communication. RuBee represents a robust and reliable technology for specific use cases, particularly in challenging environments where traditional wireless technologies may struggle to perform effectively [157], [158].
- Cellular networks offer a direct and promising avenue for smart wearable WBAN devices to connect without the need for additional bridging mechanisms. While previous network technologies were more oriented toward human communication, they encountered significant challenges when applied to wearable WBAN devices, particularly as applications evolve from basic vital sign monitoring to advanced virtual or augmented reality experiences [159]. These advanced applications demand high data rates in the gigabits per second (Gbps) range, a feat currently supported only by technologies like 802.11ac with a 1Gbps data rate. Cellular networks may involve 2G-Global System for Mobile communication (GSM), Code Division Multiple Access (CDMA), 3G-Universal Mobile Telecommunications

FIGURE 8. The most common types of communication protocols that are used for PMS in healthcare systems.

System (UMTS), CDMA2000, 4G-Long Term Evolution (LTE-A), and 5G. Due to the imperative of low energy consumption, cellular wireless technology has not yet seen widespread adoption in wearable WBAN devices. However, there is optimism for the future development of cellular technologies that are more suitable for wearable and body-worn WBAN applications, potentially enabling a broader range of advanced healthcare and entertainment experiences on wearable devices. Such cellular networks enable voice and data communication over long distances. Satellite communication could also be used to provide coverage in remote areas or to extend the reach of cellular networks.

FIGURE 8 summarizes the most common communication protocols, which can be used for PMS. Table 2 demonstrates the key features of the most common types of communication protocols.

VI. CLOUD COMPUTING AND DISTRIBUTED COMPUTING

Connecting a multitude of physical objects, such as humans, animals, smartphones, and PCs equipped with sensors, and linking them to the Internet generates vast amounts of data known as "big data." Such a big data scale necessitates smart and efficient storage, processing, and retrieval mechanisms. However, traditional hardware and software tools often struggle to handle such immense data volumes within acceptable time slots. In order to accomplish this, the US National Institute of Standards and Technology (NIST) invented cloud computing, which offers a shared programmable network of networks, servers, storage, applications, and services that can be accessed on demand. A longer explanation of cloud computing can be found in the next subsection.

A. CLOUD COMPUTING

A revolutionary concept known as "cloud computing" makes it possible to lease computing resources in real time, such as processing power, storage, and networking, without requiring upfront commitments from customers [123], [160]. With cloud services, people can reliably and economically use and maintain resources from a distance [161].

СР	Standard	Data rate	Spectrum	Data	Topology	Benefits	Cost
				range			
RFID	ISO 18000 6c	424 kbps	135 kHz	>50 cm	Point to-	Non-line-of-sight,	High
	EPC		13.56 MHz	>50 cm	point Tags	Faster rates, durability,	
	Class 1 Gen2		866-960	>3 m	and Reader	improved tracking,	
			MHz	>1.5 m		enhanced security, and	
			2.4 Ghz			versatility	
NFC	ISO 14443	100 kbps -	2.45 GHz	<10 cm	point-to-	Simple and Secure, Short	High
	ECMA-340	10 Mbps			point	Range, Versatility	
	NFC Forum						
	Specifications						
ZigBee	IEEE 802.15.4	40 - 250	868/915 MHz	10-20 m	Mesh, star,	Low power and low cost,	Low
		Kbps	2.45 GHz		and cluster	reliability, scalability	
					tree		
Bluetooth	IEEE 802.15.1	1 - 24	2.4 GHz	8-10 m	Piconet	Convenience	Low
		Mbps					
UWB	IEEE 802.15.4z	50 Mbps	Wide range	30 m	Point-to-	High data rates, low	High
					point and	power consumption,	
					multi-point	low latency, Strong	
						Resistance to	
						Interference, and Secure	
						communication	
WiFi	IEEE 802.11	1 Mbps -	5-60 GHz	20-100 m	Star and	Speed, flexibility	High
	a/b/c/d/g/n	6.75 Gbps			point-to-		
					point		

TABLE 2.	Summary of the k	ey features of the most	common types of	communication	protocols.
----------	------------------	-------------------------	-----------------	---------------	------------

For healthcare applications based on IoT, which involves numerous embedded devices like sensors and actuators generating big data, cloud computing emerges as an ideal solution for storing and processing this data efficiently. Cloud computing offers a robust platform for storing, managing, and analyzing vast volumes of healthcare data produced by IoT devices [162], [163], [164]. Cloud infrastructure offers the scalability and computational power needed to manage and analyze big data generated by IoT devices, facilitating complex computations to extract actionable insights [165]. Cloud computing utilizes computing resources, networking capabilities, storage, and other essential elements critical for operational efficiency without requiring substantial capital investments. This technology provides rapid elasticity, selfhealing, self-configuration, and ubiquitous access, offering significant benefits. Cloud services offer the illusion of infinite computing resources on demand, delivering various services such as Infrastructure as a Service (IaaS), Platform as a Service (PaaS), Database-as-a-Service (DBaaS) and Software as a Service (SaaS). Both public and private clouds can be efficiently used where the public is accessible to the general public and private clouds are dedicated to single organizations, while hybrid solutions integrate both models. Cloud computing in the healthcare sector presents compelling opportunities to contain integration costs, optimize resources, simplify processes, and enhance service quality [166], [167], [168]. Cloud computing serves as a foundational technology

for the IoT, enabling efficient processing and analysis of vast amounts of sensor-generated data within the IoT architecture. With the use of cloud based technologies, healthcare systems can address the challenges posed by increasing digital data, improve service outcomes for patients, and streamline health information management. The adoption of cloud computing in healthcare, often referred to as Healthcare as a Service (HaaS) [169], is considered to revolutionize health information technology, benefiting healthcare research and service delivery. Cloud computing offers a valuable infrastructure for

storing and managing patient medical data efficiently, especially given the volume of data generated per patient, which can overwhelm traditional storage systems. The use of cloud-based solutions helps address the challenges of handling large-scale healthcare data by leveraging technologies like big data analytics to extract valuable insights and present information in a more accessible format for healthcare professionals [131]. This approach not only improves data management and accessibility but also supports advanced analytics, facilitating more informed decision-making and enhancing patient care. By utilizing cloud-based infrastructure, healthcare providers can efficiently manage and analyze vast amounts of patient data, ultimately leading to improved patient monitoring and personalized treatment strategies. The cloud also act as on on-demand service that is very cost-efficient solution to medical services [22].

Cloud computing offers virtually limitless storage and computing resources in data centers, empowering healthcare stakeholders to derive valuable insights from analyzed and stored data. Physicians can remotely monitor patient health status using data collected from various sensors stored in the cloud, enabling timely interventions and treatment recommendations. Despite its advantages, cloud computing introduces latency challenges due to data transmission over the internet for analysis and retrieval, which may not be suitable for emergency healthcare services. To address latency issues, distributed computing such as fog/edge computing is emerging as complementary solutions in IoT healthcare applications, optimizing data processing and response times at the network edge. The work in [170] presented a net architecture for resource preservation that combines cloud and edge computing, with an emphasis on maximising important performance metrics like average patient waiting time, length of stay for patients, and resource utilisation rate. This framework emphasizes high reliability, efficiency, and security in healthcare systems. By leveraging cloud and edge computing capabilities, the proposed framework aims to model and optimize critical metrics to enhance overall system performance and patient outcomes. The integration of cloud and edge computing allows for efficient resource utilization, improved patient management, and enhanced security measures, contributing to the development of more reliable and effective healthcare systems. Through their work, Oueida et al. highlight the potential of cloud-edge integration in addressing key challenges and optimizing healthcare operations, paving the way for advanced frameworks that prioritize performance, reliability, and security in healthcare environments. In the following subsection, we will discuss fog computing and mobile edge computing (MEC).

B. DISTRIBUTED COMPUTING

Traditional cloud-centric architectures provide a number of issues that call for distributed computing paradigms like fog computing and MEC, especially in the context of healthcare applications. Using centralised cloud resources grows more and more wasteful as the amount of data produced by IoT sensors and devices keeps growing. In particular, conventional cloud architectures may struggle to meet the latency requirements of real-time healthcare applications, where rapid data processing and response times are essential for PMS. Furthermore, the geographic distribution of healthcare services and the expansion of IoT devices raise the demand for enabling distributed computing solutions that can bring computation and data storage closer to the point of data generation. Fog computing and MEC offer decentralized approaches that enable faster data processing, reduced latency, improved scalability, and enhanced privacy and security for healthcare applications [39]. These distributed computing models facilitate efficient utilization of resources, optimize network bandwidth, and support seamless integration with existing healthcare infrastructure,

ultimately enhancing the performance and reliability of healthcare systems.

Fog computing has recently been proposed by Cisco [171], [172]. Fog computing, sometimes referred to as cloudlet or edge computing, serves as a link between smart devices and extensive cloud computing and storage services. [130]. Fog computing can be implemented in gateways or edge devices, allowing for real-time data processing, local decisionmaking, and faster response times. By processing data nearer the site of data production within the network edge, fog computing expands on the idea of cloud computing. This local processing approach reduces data traffic, distance, and latency, which is crucial in latency-sensitive healthcare applications. At the network edge, data is initially gathered through an IoT network and then processed locally using edge devices. Once local processing is complete, the data is forwarded to the cloud for further computational operations and memory storage. Fog computing can utilize various communication protocols such as Wi-Fi, Zigbee, Bluetooth, and many other communication protocols. By enabling customisable decentralised networks, fog computing can help to manage and process data at the network's edge more effectively. For delay-sensitive applications like PMS, where real-time data processing and minimal latency are essential for efficient operation, fog computing is very helpful. Healthcare systems rely heavily on fog computing because it provides substantial benefits for real-time data processing [173], [174], [175]. Data processing may now be done close to sensors thanks to fog computing, which frees it from the cloud. Fog computing specifically makes use of sensors, network gateways, or local nodes to store and perform initial data processing. This would enable doctors and other clinicians to deliver emergency medical care and would allow for a significant reduction in latency.

Data storage within fog nodes allows for data replication or segregation, and security features ensure data confidentiality, integrity, and privacy. Fog computing defines the functionality of edge computing and makes processing, storage, and networking between end devices and cloud data centres easier. It operates at the perception and networking layers of the Internet of Things architecture. Fog nodes empower end-user devices to collaborate in tasks involving storage, management, and network communication, reducing latency by executing these tasks near or at the end user's location. Similar to the cloud, a fog server can store various content types, including health data, videos, audio, and local information like maps and nearby restaurant availability [176]. This architecture, depicted in typical fog computing diagrams, is well-suited for IoT applications, where the proliferation of communication-capable objects and smart devices demands efficient processing at the network's edge. Fog computing enables higher service rates, and better Quality of Service (QoS), and reduces the burden on the cloud, making it an ideal solution for IT convergence scenarios [176].

Fog computing offers several distinctive features that enhance its utility within IoT systems. Positioned between smart objects and cloud data centers, fog resources significantly improve delay performance by reducing the distance data must travel. This architecture is characterized by the deployment of numerous "micro" centers closer to end-users, leveraging their cost-effectiveness compared to expansive cloud data centers. As IoT systems scale with increasing end-users, fog computing allows for seamless scalability through the deployment of additional micro fog centers, a feat not feasible with traditional cloud data centers due to cost constraints. Fog computing also ensures resilient and replicated services, supporting a dense array of devices with mobility. With its proximity to end-users, fog resources facilitate real-time interactive services and enable on-thefly data analysis by aggregating and processing data before forwarding it to cloud data centers for further refinement, demonstrating interoperability with various cloud providers and offering enhanced performance for critical applications. This decentralized approach to computing optimizes efficiency and responsiveness within IoT ecosystems.

A fog node provides network-wide monitoring, preprocessing, storage, and security features. While preprocessing entails fundamental data analysis required for emergency healthcare services, monitoring capabilities allow the node to track resources and services. To optimize computation and energy usage without compromising functionality, models are developed specifically for edge-dependent healthcare sectors. The fog computing approach is an effective technique that integrates agile computing into PMS. This approach involves distributing computations between the cloud and edge devices using a hierarchical structure, leveraging the advantages of both edge and cloud environments for healthcare data analysis from IoT devices [177]. Healthcare applications that require quick response times can be implemented with the help of edge computing integration, and cloud computing offers plenty of processing and memory capacity. Fog computing and the cloud work together to improve performance in the healthcare industry.

The necessary tasks associated with edge computing can be completed using a virtualized software-based platform that leverages software-defined networks (SDN) and network functions virtualization (NFV) to provide a flexible, scalable, and effective network architecture [178], [179], [180]. With NFV, a single-edge computing server can offer computing services to several mobile devices. This is accomplished by building various virtual computers with the capacity to perform numerous activities at once. Furthermore, network function virtualization (NFV) has promise for managing computer services and improving the scalability, flexibility, and dependability of network services [179].

Mobile edge computing (MEC) is considered a key technology in the 5G network and is defined as an essential form of edge computing, which operates at the edge of cellular networks, enabling rapid processing of large datasets upon reception [181], [182]. MEC involves decentralizing service-specific processing, computing, and data storage from the centralized cloud core network to edge network nodes situated near data sources and endusers [172], [183]. Pushing mobile computing, network control, and storage to the network edges (base stations and access points, for example) is the primary function of MEC. This allows resource-constrained mobile devices to run computation-intensive and latency-critical applications [122]. By hosting compute-intensive apps, handling massive amounts of data before transferring it to the cloud, delivering cloud computing capabilities inside the radio access network (RAN) near mobile users, and providing context-aware services, MEC can maximise the use of mobile resources [184]. MEC has shown to have enormous potential for improving healthcare systems' performance, allowing mobile PMS in those systems. The MEC can provide a number of higher-level functions in these systems, including embedded data mining, local storage, real-time local data processing, and data transfer control [185].

By leveraging a network of servers, MEC can offload intensive computation tasks, contributing to ultra-reliable low-latency communication (URLLC), and traffic optimization [122], [183]. Implementing MEC in PMS promises significant benefits, including increased network capacity, efficient handling of delay-sensitive tasks, and offloading of network-intensive computations to reduce energy consumption and prolong battery life in the devices [186]. MEC plays a vital role in enhancing network capabilities, optimizing services, managing energy and resources efficiently, and reducing the burden on backhaul infrastructure [182]. By providing real-time data processing and analytics at the network edge, MEC integration with PMS may increase overall efficiency in healthcare applications by reducing latency and enabling faster response times.

MEC represents a significant advancement in addressing the challenges associated with managing remote devices, particularly concerning security, reliability, latency, and energy efficiency issues in healthcare systems. Recent research on EdgeCare presented in [187], emphasizes the importance of security in mobile healthcare systems. With strong security features integrated, EdgeCare offers a safe and effective data management solution that handles medical data and enables data trading at edge servers. Similarly, The work in [170] proposed BodyEdge, which introduces innovative body healthcare architectures featuring tiny mobile client modules and edge gateways. The proposed architectures enable the collection and local processing of data from various scenarios, focusing on improving latency and reliability performance within healthcare applications. Additionally, the work in [188] has implemented accurate and lightweight classification mechanisms using edge computing to detect seizures at the network edge based on vital sign information. The research findings demonstrate substantial performance improvements over conventional non-MEC

remote monitoring systems, achieving high classification accuracy for seizures, extending battery life, and significantly reducing transmission delays. Through initiatives like EdgeCare and BodyEdge, MEC demonstrates its potential to revolutionize data management and processing, offering secure, efficient, and reliable solutions at the network edge. These advancements highlight the transformative impact of MEC in enabling real-time data processing, reducing latency, and improving overall system responsiveness. As researchers continue to explore the potential of MEC, incorporating security considerations and innovative architectures, the field is poised to advance further, addressing critical challenges in remote device management and facilitating the development of advanced applications with enhanced performance and reliability.

The MEC network can dynamically handle various services thanks to SDN. Additionally, it could separate the control and data planes [189], [190]. For an edge computing network, SDN can offer a logically centralised control platform [191]. By using such a centralised control platform, the network's scalability problem could be addressed, optimal resource allocation could be achieved, overhead could be decreased, dependability could be increased, and most crucially, latency could be decreased.

VII. APPLICATION LAYER AND PMS BENEFITS

The application/action layer in PMS serves as the central hub, typically represented by the medical server. Currently, most IoT services have application and user interfaces. This layer plays a pivotal role in processing and presenting data generated by IoT devices to healthcare providers or integrating it seamlessly into electronic health records (EHR) systems. This layer provides data control services. It can mostly be affected by the type of data traveling through the IoT network [133]. Within this layer, various functionalities are enabled to support visualization, alerts, and decision-making tools for clinicians. Moreover, the application layer extends beyond mere data presentation, incorporating advanced features like self-management, active assisted living (AAL), behavioral analysis, big data analytics, emergency detection, disease prevention strategies, and mobile health (m-Health) initiatives [68]. These aspects collectively enhance the system's capability to provide efficient healthcare services.

Through the application layer, clinicians gain access to intuitive interfaces that facilitate real-time visualization of patient data and enable timely intervention when abnormalities or critical events are detected. This layer also fosters proactive healthcare management through the implementation of self-management tools, empowering patients to monitor and manage their conditions effectively. Additionally, the integration of big data analytics within this layer enables deep insights into patient trends and population health, facilitating targeted interventions and personalized care strategies. Furthermore, the integration of emergency detection and disease prevention mechanisms underscores the proactive nature of modern patient monitoring systems, aiming to mitigate risks and enhance overall patient outcomes. This convergence of technologies underpins the evolution of electronic health (e-Health) and m-Health paradigms [69], reshaping the landscape of healthcare delivery towards more connected, data-driven, and patient-centric models.

The essential task of PMS is to monitor patients in different places such as home, office, hospital, battlefield, and other places. Using PMS can prevent transmissible and infectious diseases, keep the patient's life safer, and provide quick diagnostics and treatment in addition to other health assistance tools. In addition, the economic cost, morbidity, and mortality associated with human beings have been reduced using remote PMS, which appears as a promising solution and can succeed on a large scale. Furthermore, one key area where PMS proves invaluable is in hospital environments. In such environments monitoring patient's vital signs can improve patient safety and enable early detection of deteriorating conditions. In intensive care units (ICUs), PMS plays a crucial role in providing real-time data to healthcare providers, facilitating prompt interventions, and improving patient outcomes. Beyond hospitals, PMS extends its utility to home care settings, allowing patients with chronic illnesses or post-surgery recovery needs to be monitored remotely. This remote monitoring capability promotes patient independence while providing clinicians with vital data for timely interventions. Moreover, PMS aids in the management of elderly patients, ensuring their well-being by tracking vital signs and detecting emergencies. Nurses can benefit significantly from PMS to monitor patients remotely, allowing for proactive interventions and reducing the need for frequent clinic visits. This approach not only improves patient comfort but also optimizes healthcare resource utilization. In emergency situations, PMS enables real-time monitoring of critical patient data, providing crucial information to emergency medical personnel before arrival at the hospital. This capability enhances triage and treatment decisions, potentially saving lives. Additionally, PMS plays a vital role in ensuring the security and integrity of medical data. By employing robust data encryption and secure transmission protocols, PMS can allow safe sensitive health information against unauthorized access or breaches. This aspect is critical in maintaining patient confidentiality and compliance with data protection regulations. Furthermore, PMS supports seamless integration with EHRs, enabling healthcare providers to access comprehensive patient histories and monitor trends over time. This accessibility enhances clinical decision-making and fosters continuity of care across different healthcare settings, ultimately improving healthcare systems.

PMS plays a vital role in ambient assisted living (AAL) environments by enabling continuous remote monitoring of individuals with chronic illnesses or disabilities. In AAL settings, PMS empowers individuals to live independently while providing caregivers and healthcare professionals with real-time data on vital signs and activities. This proactive

monitoring allows for early detection of health issues or emergencies, enhancing overall safety and well-being within the home environment. PMS in AAL environments facilitates personalized care plans based on individual health trends and alerts caregivers to potential concerns promptly. By integrating with smart home systems and wearable devices, PMS supports seamless data collection and analysis, contributing to improved health outcomes and quality of life for individuals. Applications for AAL track and assess a range of fundamental daily tasks, such eating, and cleaning. They give the subject suitable assistance involving carers and permit the subject to live independently. Fall detection and movement monitoring systems identify ambulatory behaviours such as unintentional falls, static postures, dynamic activities, and location tracking. Real-time applications are used by physiological PMS to monitor and diagnose vital signs in dependent and chronically unwell people, including those with diabetes, hypertension, and cardiovascular disorders [73].

VIII. RELATED CHALLENGES AND FUTURE RESEARCH DIRECTIONS IN PMS BASED ON IOT IN HEALTHCARE SYSTEMS

Despite the numerous benefits offered by PMS, which were discussed above, there remain several significant challenges that need to be addressed. One key challenge is related to data confidentially, as PMS deals with sensitive medical information that must be protected. Ensuring the confidentiality, integrity, and availability of patient data in PMS is crucial but can be complex due to evolving cybersecurity threats. Several works have been proposed to address this issue in the healthcare domain and PMS networks, see, e.g., [192], [193], [194].

Noting that the interoperability is considered as critical challenge in integrating new IoT-based PMSs with existing healthcare infrastructures, as healthcare facilities often use a variety of monitoring devices that may not seamlessly communicate with each other. As IoT devices generate vast amounts of real-time health data, seamless communication and data exchange between these devices and traditional health systems become essential. However, the lack of standardized protocols and varying data formats across different platforms create significant barriers. Ensuring that IoT-based PMSs can effectively communicate with EHRs and other legacy systems requires overcoming these interoperability issues, which is vital for obtaining an efficient healthcare system.

Furthermore, the implementation of PMS systems in healthcare presents several significant challenges that must be carefully investigated in future to ensure the successful implementation of PMS in healthcare system. The issues of informed consent and ethical considerations need to considered. To this end, patients must be thoroughly informed about how their health data will be collected, used, and protected, emphasizing the importance of maintaining their autonomy and privacy. This transparency is crucial to build trust between patients and healthcare providers. In addition, integrating PMS systems into existing healthcare workflows requires substantial changes in how hospital staff operate. This transition often demands extensive training and adaptation. The challenge is to seamlessly incorporate PMS technologies without disrupting the delivery of care. Further investigation in this regard is needed. Moreover, the financial aspects cannot be overlooked; the high initial costs associated with implementing PMS systems, coupled with the uncertain landscape of insurance reimbursement, can be significant challenges for healthcare institutions considering adoption. These financial hurdles, along with the need for clear reimbursement policies, add another layer of complexity to the widespread integration of PMS technologies. Healthcare regulators and policymakers need to create clear, consistent reimbursement frameworks for PMS technologies. This includes establishing guidelines that recognize the value of remote monitoring and IoT-based healthcare services, ensuring that these services are reimbursed similarly to traditional in-person care. Besides, educating healthcare providers, patients, and policymakers about the benefits and cost-effectiveness of PMS technologies is essential. Advocacy efforts can help build support for the adoption of these technologies and for the development of reimbursement models that ensure sustainability.

Additionally, scalability and cost-effectiveness pose challenges, particularly for smaller healthcare providers who may struggle to implement and maintain sophisticated PMS solutions. Overcoming these challenges requires concerted efforts in technology development, standardization, and ongoing research works. Moreover, the major components of PMS must be selected carefully. Examples of these components are the types of sensors, processing methods, contact methods, and communication networks. In addition, in large-scale PMS, there is a large amount of data collected from the sensors. In summary, the advancement in wearable devices, which aim to achieve low-cost computing, big storage devices, low power consumption, and the development in communications technologies are paving the way for a low-cost, applicable, and effective PMS. In the following, we provide some of the related challenges in PMS-based IoT in healthcare systems.

• Data and resource management: The IoT-based healthcare systems consist of numerous interconnected devices that generate massive amounts of data, which must be assimilated, stored, and analyzed in order to be analyzed effectively. IoT devices and sensors will generate different types of data; for instance, medical devices generate image data, whereas other devices generate video data. This gives rise to well-known big data issues. High processing power and large data storage capacity are needed to handle this healthcare data. Therefore, PMS based on IoT with healthcare systems will require advanced data-processing mechanisms due to the varying performance requirements for devices and applications. Besides, advanced algorithms for processing the data in edge computing or cloud computing might be helpful in addressing the big data issue. Therefore, further research is needed to investigate data management issues in PMS. Sensor nodes are constrained in terms of store capacity and processing power. Therefore, efficiently managing the resources in PMS PMS-based IoT healthcare systems is essential. Further research regarding the resource management of PMS is required.

Security and privacy: Due to the critical sensitivity of the data processed in the PMS, it is essential to secure and maintain constant connectivity between medical devices. In addition, patient privacy and security concerns are becoming increasingly important as the number of IoT devices in healthcare systems grows at an exponential rate. In order to handle the plethora of medical assets employed throughout the healthcare system, a number of operational and interoperability difficulties must also be addressed. One such challenge is the integration of varied technologies within the IoT ecosystem. Note that measurement data are gathered from various instruments, sensors, and devices and sent via gateways to an Internet-connected server housing databases. Hence, ensuring reliable communication is essential. Furthermore, security requirements for PMS include availability, confidentiality, integrity, access control, and authentication. These requirements are essential for the medical data shared within hospital networks. Furthermore, as the quantity of IoT devices in healthcare systems increases at an exponential rate, issues about patient privacy and security are becoming more and more significant. Many operational and interoperability challenges need to be resolved in order to manage the abundance of medical assets used across the healthcare system. The IoT ecosystem presents several challenges, one of which is the integration of diverse technologies. Furthermore, the use of wireless technology in healthcare systems without taking security precautions leaves users open to privacy problems. To this end, healthcare providers should consider privacy. Traditional security methods based on existing cryptography solutions face challenges when applied to IoT-based healthcare applications due to resource limitations, specific IoT architecture requirements, and the high-security standards demanded. Physical security, authentication, network security, computer security, and storage security are all included in data security. The utilization of traditional cryptographic approaches becomes impractical within IoT-based healthcare applications. Additionally, IoT gateways often prioritize basic functions over critical security measures like authorization and authentication, highlighting further complexities in ensuring robust security within these systems. We need to design a PMS model that guarantees that only authorized individuals have an access to a patient's medical data. In brief, maintaining security and privacy are essential for PMS, encompassing numerous elements and procedures such as data gathering, communication, sensor and device use. Therefore, to protect sensitive data generated by devices and personal information, advanced technologies are needed. To solve the security and privacy concerns in PMS, more research is required to examine the application of ML and AI approaches.

- Managing heterogeneity in data: Numerous data can be generated by health applications. These data are obtained from a range of heterogeneous sources. Heterogeneity in PMS is mostly encountered in two forms: sensor heterogeneity and data heterogeneity. Multimodal sensors are used to gather data, and they are diverse in terms of organization, format, and meaning. Consequently, datasets are challenging to distribute and reuse since there aren't any official summaries. Despite the efforts has been dedicated to create sensors when describing semantics with data models, it is necessary to standardize the modeling of sensor data to reflect heterogeneous sources of data. Therefore, advanced technologies are required to handle such data heterogeneity. The best possible way to address the data heterogeneity is to make use of advanced artificial intelligence algorithms. Consequently, more study is needed to determine how artificial intelligence and machine learning approaches might be used to manage the variability of PMS data.
- Data integration: Data from multiple sources must be integrated. Scales, social networks, imaging systems, blood pressure monitors, glucose metres, heart rate monitors, fitness equipment, blood oxygen monitors, and many more internet sites are some examples of these sources [195]. Valid and significant results cannot be obtained from the collection of data from diverse sources unless the syntax, structure, and meaning of the sentences are correctly understood. An accurate comprehension can create intelligent applications or a blending procedure. Creating efficient data integration and analytics platforms for clinical decision-making can consolidate disparate data sources into comprehensive and unbiased analyses, offering rapid and insightful solutions that are not achievable through manual processes. Developing such systems for clinical decision support can convert static and offline datadriven guidelines, which are constantly evolving, into dynamic and interactive algorithms accessible online for immediate execution. Integrating patient medical reports into these systems can facilitate the creation of point-of-care decision-making tools. These analytical tools can enhance healthcare response capabilities by enabling epidemic surveillance, geospatial analysis, cluster outbreak reporting, and the development of accurate therapeutic algorithms to address global health challenges, such as pandemics [196]. A heterogeneous network contains a range of sensors with different

sensing techniques or communication spans. Devices from various vendors will be present in a network, and communication between those devices must be guaranteed. Furthermore, interoperability is necessary to ensure that a system can connect to another network that performs different functions. To process the data and make decisions, the data structure must be consistent. Further research investigations to address the aforementioned challenges are required.

- Context-awareness: Applications, interfaces, and services tailored to the user's situation and its changing needs are presented in a relevant, appropriate, and personalised manner by a health context-awareness system. A thorough understanding of a subject's surroundings enables the system to ascertain the subject's actual help needs on the fly. In PMS, obtaining a high-level coherent abstraction of contextual data requires accurate interpretation of data and events. Apart from traditional monitoring methods, it is important to reliably identify emergency scenarios and declines in health. The two main obstacles to the creation of context-aware medical apps are, generally speaking, data acquisition (the amount of information gathered) and data analysis methodologies, which include presenting context-relevant services and information. Contextawareness interpretation and data representation are critical components that impact a health surveillance system's effectiveness. Rapid intervention is made easier by accurate emergency situation identification, which raises the standard of healthcare. Although many current systems fall short of achieving a high degree of contextual awareness, gathering comprehensive and all-encompassing contextual data continues to be a crucial unresolved issue that required further focus. A context-sensitive monitoring system's design must take behavioural, physiological, and environmental factors into account when observing, interpreting, and analysing patient situations. All pertinent context factors, which primarily include place, time, objects, position, frequency, and human activity, must also be considered by the system. System performance is also influenced by historical data, which must be considered. This includes health data (such as diagnoses, illnesses, and treatments), daily behaviour and past changes, and environmental variables (such as humidity and temperature). However, the same context data needs to be accessible across several distribution components that employ various network protocols. This allows different carers to access the contextual data that they require from the infrastructure via the network that is available, which can frequently change in a large setting.
- Power consumption: Another significant challenge that faces sensors in general, particularly in BSN, is energy consumption. The sensors' short battery life may have an adverse effect on the PMS. For monitored subjects,

battery replacement and charging in the majority of BAN systems currently in use are inefficient and timeconsuming, particularly in architectures with multiple sensors. Any communication device's battery life is typically influenced by its duty cycle, transmission range, and communication channel usage. Sensor nodes can periodically turn On/Off their radio interface with a centrally assigned time slot using the widely used to save energy. Besides, advanced energy harvesting and wireless charging techniques for PMS need to be investigated. Furthermore, low power consumption can also be achieved by applying smart processing algorithms. Consequently, further real-world research is needed to determine how well energy is transferred and harvested in PMS. Additionally, the circuit complexity of the devices may be raised to enable efficient energy harvesting design in PMS. To fully realise the benefits of energy harvesting in PMS, this can also be seen as one of the technological difficulties that need to be resolved.

- Scalability: The healthcare system's scalability is still a problem today. Billions of IoT devices on the network will generate vast amounts of health data. In a similar vein, the amount of data that must be handled and preserved will increase rapidly. This will present a massive data challenge for the healthcare industry. A scalable system is needed to store and analyse this data from IoT devices in order to solve this problem. On the future, data collected from linked IoT devices will need to be stored on the cloud using big data analytics and effective edge computing approaches. Plans for therapy will be improved with the use of this data. In this aspect, more research is necessary. It is imperative that big data analytics be integrated with Internet of Things-based healthcare solutions. Modern medical telematics and informatics have emerged as a result of this convergence, bringing about developments that are highly helpful for PMS sufferers, such as disease diagnosis, remote real-time health monitoring, preventive measures, and medical emergency warning systems. Consequently, further research on big data analytics in the context of healthcare is needed.
- Safety issue: Particularly when carers are absent, PMS should keep offering care services like accident detection and emergency assistance. In a variety of situations, PMS must guarantee the security of participants and carers, including participants with mental illnesses. Simple presence sensors to sophisticated video surveillance systems with real-time alerts and the ability to detect and identify patterns of movement are just a few of the tools that PMS may rely on. Therefore, especially in healthcare facilities, biometric sensors can be utilised to identify aggressive and stressed behaviours. We think that other crucial aspects that must be taken into account are the safety, acceptability, and usability of PMS. Thus, effective video surveillance systems for PMS must be created.

- Data processing: Big data is the massive amount of data generated by a large number of connected IoT devices. Millions of consumer devices have interconnected sensors and actuators that interact, collect, and transmit data. Every gadget generates or detects vast amounts of data that need additional processing. Making sure that these various data types are handled appropriately and effectively will be the main challenge, especially with a limited amount of time, processing power, and computational resources available. It was estimated that by 2020, the amount of data stored would have increased from 80,000 petabytes in 2000 to 35 zettabytes. Due to various data structures, only 20% of these are examined using conventional methods; the remaining 80% are not used in decision-making. It is anticipated that as 6G is evaluated, the amount of data will grow even more. With the aid of artificial intelligence, data-driven techniques can be created to manage big data from numerous IoT devices. Further research investigation is needed.
- Device design issue: The healthcare sector makes use of PMS-based IoT devices, which have tiny sensors, minimal processing power, little storage, and short battery lives. The internet is connected to IoT devices. Wearable technology requires network connectivity in order to offer carers access to health information. Developing IoT devices with more processing power, storage capacity, long battery life, and security that comply with mobility standards remains a research problem to this day. In this area, more research and investigation are needed.
- Monitoring accuracy: Because performing different everyday activities requires a high level of skill, monitoring and evaluating human subjects-especially the elderly-is a challenging endeavour. Subject to subject, the behaviour is arbitrary and varies. The majority of recent research on activity monitoring focuses on short-term, relatively easy tasks that are completed in a lab setting. In the actual world, however, it is evident that managing complicated and long-term tasks in terms of gathering and evaluating pertinent data is more challenging. Activities that involve several subactions can also be completed in various ways and in different sequences. In intelligent homes, long-term monitoring is helpful in identifying actual behavioural patterns. Data related to single-user actions are gathered and processed in an intelligent environment in current research experiments. In actuality, though, several topics can share one space and be completed concurrently with daily tasks. Moreover, the majority of research in this area focuses on activity recognition rather than evaluating the accuracy with which participants carry out these tasks. These issues indicate that the precision and reliability of the outcomes are still somewhat disconnected from the unpredictable and enigmatic real world of human behaviour. We think that further work has to be done to enhance the planning and creation

of monitoring systems for senior citizens in intelligent settings. The right sensors must be chosen with care in such systems and solutions in order to collect rich and pertinent data for one or more topics. For the system to successfully comprehend complicated and constantly evolving human behaviour, it must choose the most suitable approaches and strategies. To make sure that these systems are effective in terms of analysis, accuracy, and adaptability to the suggested monitoring, extensive testing is required.

- Trustworthiness: Because healthcare providers utilise data to inform decisions about patient care, data must be thorough, accurate, and consistent in order for it to be considered trustworthy. Specifically, the data produced by sensors and wearable technology is used by carers to inform their decisions and treatment plans. On the other hand, data corruption might result from unapproved exposure and disclosure, which can cause the loss of personal data. Furthermore, malware or viruses could contaminate or corrupt the medical data during transmission. Investigating the dependability of trust management on PMS requires making use of blockchain technology's features, which include integrity, immutability, and trustworthiness. The possible catastrophic repercussions of medical device trustworthiness necessitate prompt and effective resolution of any problems [197]. Traditionally, generalpurpose and embedded computer systems have used information security and fault tolerance techniques like cryptography and redundancy. However, because of their significant size and power limitations, they cannot be utilised to many wearable and implantable medical devices [197]. Therefore, creating effective and reliable security methods to ensure trustworthiness is one of the major issues that need to be considered in future research.
- Data transmission delay: A common issue in PMS development is deciding how to transfer sensor data to back-end servers for processing and analysis. Data transmissions in networking are divided into four categories: multicast, broadcast, unicast, and anycast. Multicast or broadcast schemes are frequently used by existing PMS to improve data transmission efficiency and reliability. Both schemes send packets to several recipients at once. As a result, a high volume of network traffic and transmission delay result from frequent transmissions. With the least amount of traffic overhead, the unicast scheme delivers packets to a single recipient; however, additional steps must be taken to locate a backup receiver in the event of a transmission failure. Anycast communication is a brand-new routing protocol the point at which data packets are sent to the closest recipient. This scheme is more dependable than broadcast and multicast and has less traffic overhead. Unicast is used to discover new receivers. But anycast raises the devices' complexity

FIGURE 9. The research challenges related to PMS in healthcare systems.

and network routing utilized by the plan. This challenge raises some more general issues. Hence, analyzing the transmission technologies, as well as the quantity of data sent, standard packet size, and transmission frequencies, need to be considered. Therefore, developing new PMS models that take into account the data transmission issue is of interest.

• Availability and reliability: Providing timely access to pertinent health data is another essential challenge for PMS. The dependability of data distribution may have an impact on the accessibility of health information. Numerous factors can affect the reliability of data on wireless networks in health infrastructures. These variables include routing protocols, device or network faults, power availability, network coverage, and device range. Further investigation in this regard is required.

The research challenges presented in this paper can be summarized in FIGURE 9.

IX. CONCLUSION

This paper presented a comprehensive review of PMS in healthcare systems. To this end, an in-depth study of recent trends and the current state-of-the-art research in PMS based on IoT technologies, which are essentially needed to achieve efficient healthcare services, was provided. In particular, this paper provided the effective architecture of PMS and the essential communication technologies that are required for developing a realistic PMS. Moreover, this paper emphasizes the importance and necessity of implementing IoT-based PMS in healthcare systems and discusses different types of current monitoring and tracking systems. The communication networks and protocols that are required to endure efficient healthcare systems were explored. Besides, the key fundamental benefit of PMS has been described. In addition, this report outlined a number of unresolved research issues and suggested fresh lines of inquiry for further investigation. This can benefit both academic and industrial sectors in remote healthcare applications. This paper is essential for understanding the PMS that can be used for enhancing medical services, health management, and response strategies during public health emergencies and routine care scenarios. This paper offered a comprehensive understanding of PMS, which is crucial for business and scholarly academics alike.

ACKNOWLEDGMENT

The authors would like to acknowledge the University of Baghdad, the University of Sharjah, Al-Farahidi University, and Liverpool John Moores University, for their general support.

REFERENCES

- A. Zanella, N. Bui, A. Castellani, L. Vangelista, and M. Zorzi, "Internet of Things for smart cities," *IEEE Internet Things J.*, vol. 1, no. 1, pp. 22–32, Feb. 2014.
- [2] R. Minerva, A. Biru, and D. Rotondi, "Towards a definition of the Internet of Things (IoT)," *IEEE Internet Initiative*, vol. 1, pp. 1–86, 2015.
- [3] K. K. Patel, S. M. Patel, and P. Scholar, "Internet of Things-IoT: Definition, characteristics, architecture, enabling technologies, application & future challenges," *Int. J. Eng. Sci. Comput.*, vol. 6, no. 5, pp. 1–10, 2016.
- [4] K. Takleef, K. Ali, M. A. Salim, and M. Wadi, "An overview of patient's health status monitoring system based on Internet of Things (IoT)," *Wireless Pers. Commun.*, vol. 114, no. 3, pp. 2235–2262, 2020.
- [5] R. Mieronkoski, I. Azimi, A. M. Rahmani, R. Aantaa, V. Terävä, P. Liljeberg, and S. Salanterä, "The Internet of Things for basic nursing care—A scoping review," *Int. J. Nursing Stud.*, vol. 69, pp. 78–90, Apr. 2017.
- [6] P. K. Sadhu, V. P. Yanambaka, and A. Abdelgawad, "Internet of Things: Security and solutions survey," *Sensors*, vol. 22, no. 19, p. 7433, Sep. 2022.
- [7] J. Gómez, B. Oviedo, and E. Zhuma, "Patient monitoring system based on Internet of Things," *Proc. Comput. Sci.*, vol. 83, pp. 90–97, Jan. 2016.
- [8] M. A. Khan, I. U. Din, B.-S. Kim, and A. Almogren, "Visualization of remote patient monitoring system based on Internet of Medical Things," *Sustainability*, vol. 15, no. 10, p. 8120, May 2023.
- [9] H. H. Alshammari, "The Internet of Things healthcare monitoring system based on MQTT protocol," *Alexandria Eng. J.*, vol. 69, pp. 275–287, Apr. 2023.
- [10] T. Han, K. Muhammad, T. Hussain, J. Lloret, and S. W. Baik, "An efficient deep learning framework for intelligent energy management in IoT networks," *IEEE Internet Things J.*, vol. 8, no. 5, pp. 3170–3179, Mar. 2021.
- [11] Z. Mohammadzadeh, H. R. Saeidnia, A. Lotfata, M. Hassanzadeh, and N. Ghiasi, "Smart city healthcare delivery innovations: A systematic review of essential technologies and indicators for developing nations," *BMC Health Services Res.*, vol. 23, no. 1, p. 1180, Oct. 2023.
- [12] Z. Chen, C. B. Sivaparthipan, and B. Muthu, "IoT based smart and intelligent smart city energy optimization," *Sustain. Energy Technol. Assessments*, vol. 49, Feb. 2022, Art. no. 101724.
- [13] W. A. Jabbar, T. K. Kian, R. M. Ramli, S. N. Zubir, N. S. M. Zamrizaman, M. Balfaqih, V. Shepelev, and S. Alharbi, "Design and fabrication of smart home with Internet of Things enabled automation system," *IEEE Access*, vol. 7, pp. 144059–144074, 2019.
- [14] P. Asef, R. Taheri, M. Shojafar, I. Mporas, and R. Tafazolli, "SIEMS: A secure intelligent energy management system for industrial IoT applications," *IEEE Trans. Ind. Informat.*, vol. 19, no. 1, pp. 1039–1050, Jan. 2023.
- [15] Z. Hu and H. Tang, "Design and implementation of intelligent vehicle control system based on Internet of Things and intelligent transportation," *Sci. Program.*, vol. 2022, pp. 1–11, Jan. 2022.

- [16] Statista. (2018). Statista, Internet of Things (IoT) Connected Devices Installed Base Worldwide From 2015 to 2025 (in Billions). [Online]. Available: https://www.statista.com/statistics/471264/iot-number-ofconnected-devices-worldwide/
- [17] Y. Al Mtawa, A. Haque, and B. Bitar, "The Mammoth internet: Are we ready?" *IEEE Access*, vol. 7, pp. 132894–132908, 2019.
- [18] Y. Shelke and A. Sharma, "Internet of Medical Things," Tech. Rep., 2016. [Online]. Available: https://www.aranca.com/assets/ uploads/resources/special-reports/Internet-of-Medical-Things-IoMT_ Aranca-Special-Report.pdf
- [19] precedenceresearch. (2024). Internet of Things in Healthcare Market. [Online]. Available: https://www.precedenceresearch.com/internetof-things-in-healthcare-market
- [20] S. Vishnu, S. R. J. Ramson, and R. Jegan, "Internet of Medical Things (IoMT)—An overview," in *Proc. 5th Int. Conf. Devices, Circuits Syst.* (ICDCS), Mar. 2020, pp. 101–104.
- [21] R. De Michele and M. Furini, "IoT healthcare: Benefits, issues and challenges," in Proc. 5th EAI Int. Conf. Smart Objects Technol. Social Good (GoodTechs), Sep. 2019, pp. 160–164.
- [22] A. V. L. N. Sujith, G. S. Sajja, V. Mahalakshmi, S. Nuhmani, and B. Prasanalakshmi, "Systematic review of smart health monitoring using deep learning and artificial intelligence," *Neurosci. Informat.*, vol. 2, no. 3, Sep. 2022, Art. no. 100028.
- [23] G. Muhammad, F. Alshehri, F. Karray, A. E. Saddik, M. Alsulaiman, and T. H. Falk, "A comprehensive survey on multimodal medical signals fusion for smart healthcare systems," *Inf. Fusion*, vol. 76, pp. 355–375, Dec. 2021.
- [24] S. Razdan and S. Sharma, "Internet of Medical Things (IoMT): Overview, emerging technologies, and case studies," *IETE Tech. Rev.*, vol. 39, no. 4, pp. 775–788, Jul. 2022.
- [25] G. R. Pradyumna, R. B. Hegde, K. B. Bommegowda, T. Jan, and G. R. Naik, "Empowering healthcare with IoMT: Evolution, machine learning integration, security, and interoperability challenges," *IEEE Access*, vol. 12, pp. 20603–20623, 2024.
- [26] K. N. Swaroop, K. Chandu, R. Gorrepotu, and S. Deb, "A health monitoring system for vital signs using IoT," *Internet Things*, vol. 5, pp. 116–129, Mar. 2019.
- [27] Ŷ. Yin, Y. Zeng, X. Chen, and Y. Fan, "The Internet of Things in healthcare: An overview," J. Ind. Inf. Integr., vol. 1, pp. 3–13, Mar. 2016.
- [28] R. Pratap Singh, M. Javaid, A. Haleem, R. Vaishya, and S. Ali, "Internet of Medical Things (IoMT) for orthopaedic in COVID-19 pandemic: Roles, challenges, and applications," *J. Clin. Orthopaedics Trauma*, vol. 11, no. 4, pp. 713–717, Jul. 2020.
- [29] J. V. Vaghasiya, C. C. Mayorga-Martinez, and M. Pumera, "Wearable sensors for telehealth based on emerging materials and nanoarchitectonics," *NPJ Flexible Electron.*, vol. 7, no. 1, p. 26, Jun. 2023.
- [30] M. Dhimal, F. Chirico, B. Bista, S. Sharma, B. Chalise, M. L. Dhimal, O. S. Ilesanmi, P. Trucillo, and D. Sofia, "Impact of air pollution on global burden of disease in 2019," *Processes*, vol. 9, no. 10, p. 1719, Sep. 2021.
- [31] S. S. Mishra and A. Rasool, "IoT health care monitoring and tracking: A survey," in *Proc. 3rd Int. Conf. Trends Electron. Informat. (ICOEI)*, Apr. 2019, pp. 1052–1057.
- [32] S. P. Rajan, "Review and investigations on future research directions of mobile based telecare system for cardiac surveillance," *J. Appl. Res. Technol.*, vol. 13, no. 4, pp. 454–460, Aug. 2015.
- [33] R. Negra, I. Jemili, and A. Belghith, "Wireless body area networks: Applications and technologies," *Proc. Comput. Sci.*, vol. 83, pp. 1274–1281, Jan. 2016.
- [34] Ê. Nemati, M. J. Deen, and T. Mondal, "A wireless wearable ECG sensor for long-term applications," *IEEE Commun. Mag.*, vol. 50, no. 1, pp. 36–43, Jan. 2012.
- [35] I. F. Akyildiz, A. Kak, and S. Nie, "6G and beyond: The future of wireless communications systems," *IEEE Access*, vol. 8, pp. 133995–134030, 2020.
- [36] M. Alsabah, M. A. Naser, B. M. Mahmmod, S. H. Abdulhussain, M. R. Eissa, A. Al-Baidhani, N. K. Noordin, S. M. Sait, K. A. Al-Utaibi, and F. Hashim, "6G wireless communications networks: A comprehensive survey," *IEEE Access*, vol. 9, pp. 148191–148243, 2021.
- [37] B. Cornet, H. Fang, H. Ngo, E. W. Boyer, and H. Wang, "An overview of wireless body area networks for mobile health applications," *IEEE Netw.*, vol. 36, no. 1, pp. 76–82, Jan. 2022.
- [38] M. Poongodi, A. Sharma, M. Hamdi, M. Maode, and N. Chilamkurti, "Smart healthcare in smart cities: Wireless patient monitoring system using IoT," *J. Supercomput.*, vol. 77, no. 11, pp. 12230–12255, Nov. 2021.

- [39] G. Aceto, V. Persico, and A. Pescapé, "The role of information and communication technologies in healthcare: Taxonomies, perspectives, and challenges," *J. Netw. Comput. Appl.*, vol. 107, pp. 125–154, Apr. 2018.
- [40] M. Aledhari, R. Razzak, B. Qolomany, A. Al-Fuqaha, and F. Saeed, "Biomedical IoT: Enabling technologies, architectural elements, challenges, and future directions," *IEEE Access*, vol. 10, pp. 31306–31339, 2022.
- [41] E. Escobar-Linero, L. Muñoz-Saavedra, F. Luna-Perejón, J. L. Sevillano, and M. Domínguez-Morales, "Wearable health devices for diagnosis support: Evolution and future tendencies," *Sensors*, vol. 23, no. 3, p. 1678, Feb. 2023.
- [42] S. Abdulmalek, A. Nasir, W. A. Jabbar, M. A. M. Almuhaya, A. K. Bairagi, M. A.-M. Khan, and S.-H. Kee, "IoT-based healthcaremonitoring system towards improving quality of life: A review," *Healthcare*, vol. 10, no. 10, p. 1993, Oct. 2022.
- [43] J. Goudge, L. Gilson, S. Russell, T. Gumede, and A. Mills, "Affordability, availability and acceptability barriers to health care for the chronically ill: Longitudinal case studies from south Africa," *BMC Health Services Res.*, vol. 9, no. 1, pp. 1–18, Dec. 2009.
- [44] M. L. Sahu, M. Atulkar, M. K. Ahirwal, and A. Ahamad, "Vital sign monitoring system for healthcare through IoT based personal service application," *Wireless Pers. Commun.*, vol. 122, no. 1, pp. 129–156, Jan. 2022.
- [45] K. Haricha, A. Khiat, Y. Issaoui, A. Bahnasse, and H. Ouajji, "Towards smart manufucturing: Implementation and benefits," *Proc. Comput. Sci.*, vol. 177, pp. 639–644, Jan. 2020.
- [46] J. M. Portnoy, A. Pandya, M. Waller, and T. Elliott, "Telemedicine and emerging technologies for health care in allergy/immunology," *J. Allergy Clin. Immunol.*, vol. 145, no. 2, pp. 445–454, 2020.
- [47] N. M. Quy, L. A. Ngoc, N. T. Ban, N. V. Hau, and V. K. Quy, "Edge computing for real-time Internet of Things applications: Future internet revolution," *Wireless Pers. Commun.*, vol. 132, no. 2, pp. 1423–1452, Sep. 2023.
- [48] T. Yesmin, M. W. Carter, and A. S. Gladman, "Internet of Things in healthcare for patient safety: An empirical study," *BMC Health Services Res.*, vol. 22, no. 1, p. 278, Mar. 2022.
- [49] A. V. Kelarev, A. Stranieri, J. L. Yearwood, and H. F. Jelinek, "Empirical study of decision trees and ensemble classifiers for monitoring of diabetes patients in pervasive healthcare," in *Proc. 15th Int. Conf. Network-Based Inf. Syst.*, Sep. 2012, pp. 441–446.
- [50] A. Papa, M. Mital, P. Pisano, and M. Del Giudice, "E-health and wellbeing monitoring using smart healthcare devices: An empirical investigation," *Technol. Forecasting Social Change*, vol. 153, Apr. 2020, Art. no. 119226.
- [51] A. Dohr, R. Modre-Opsrian, M. Drobics, D. Hayn, and G. Schreier, "The Internet of Things for ambient assisted living," in *Proc. 7th Int. Conf. Inf. Technol., New Generat.*, Apr. 2010, pp. 804–809.
- [52] J. Mohammed, C.-H. Lung, A. Ocneanu, A. Thakral, C. Jones, and A. Adler, "Internet of Things: Remote patient monitoring using web services and cloud computing," in *Proc. IEEE Int. Conf. Internet Things* (*iThings*), *IEEE Green Comput. Commun. (GreenCom) IEEE Cyber*, *Phys. Social Comput. (CPSCom)*, Sep. 2014, pp. 256–263.
- [53] M. S. Jassas, A. A. Qasem, and Q. H. Mahmoud, "A smart system connecting e-health sensors and the cloud," in *Proc. IEEE 28th Can. Conf. Electr. Comput. Eng. (CCECE)*, May 2015, pp. 712–716.
- [54] H. Mansor, M. H. A. Shukor, S. S. Meskam, N. Q. A. M. Rusli, and N. S. Zamery, "Body temperature measurement for remote health monitoring system," in *Proc. IEEE Int. Conf. Smart Instrum., Meas. Appl.* (*ICSIMA*), Nov. 2013, pp. 1–5.
- [55] K. M. Kumar and R. S. Venkatesan, "A design approach to smart health monitoring using Android mobile devices," in *Proc. IEEE Int. Conf. Adv. Commun., Control Comput. Technol.*, May 2014, pp. 1740–1744.
- [56] N. P. Jain, P. N. Jain, and T. P. Agarkar, "An embedded, GSM based, multiparameter, realtime patient monitoring system and control—An implementation for ICU patients," in *Proc. World Congr. Inf. Commun. Technol.*, Oct. 2012, pp. 987–992.
- [57] S. Roy and R. Gupta, "Short range centralized cardiac health monitoring system based on ZigBee communication," in *Proc. IEEE Global Humanitarian Technol. Conf.-South Asia Satell. (GHTC-SAS)*, Sep. 2014, pp. 177–182.
- [58] R. Piyare, "Internet of Things: Ubiquitous home control and monitoring system using Android based smart phone," *Int. J. Internet Things*, vol. 2, no. 1, pp. 5–11, 2013.

- [59] K. Malhi, S. C. Mukhopadhyay, J. Schnepper, M. Haefke, and H. Ewald, "A zigbee-based wearable physiological parameters monitoring system," *IEEE Sensors J.*, vol. 12, no. 3, pp. 423–430, Mar. 2012.
- [60] M. Ghamari, B. Janko, R. Sherratt, W. Harwin, R. Piechockic, and C. Soltanpur, "A survey on wireless body area networks for eHealthcare systems in residential environments," *Sensors*, vol. 16, no. 6, p. 831, Jun. 2016.
- [61] A. Rghioui, J. Lloret, S. Sendra, and A. Oumnad, "A smart architecture for diabetic patient monitoring using machine learning algorithms," *Healthcare*, vol. 8, no. 3, p. 348, Sep. 2020.
- [62] A. Gondalia, D. Dixit, S. Parashar, V. Raghava, A. Sengupta, and V. R. Sarobin, "IoT-based healthcare monitoring system for war soldiers using machine learning," *Proc. Comput. Sci.*, vol. 133, pp. 1005–1013, Jan. 2018.
- [63] F. Dahan, R. Alroobaea, W. Y. Alghamdi, M. K. Mohammed, F. Hajjej, D. M. Alsekait, and K. Raahemifar, "A smart IoMT based architecture for e-healthcare patient monitoring system using artificial intelligence algorithms," *Frontiers Physiol.*, vol. 14, Jan. 2023, Art. no. 1125952.
- [64] W. Huang, A. Song, B. Wan, and J. Guo, "Wearable health monitoring system based on layered 3D-mobilenet," *Proc. Comput. Sci.*, vol. 202, pp. 373–378, Jan. 2022.
- [65] I. Priyadarsini, B. Tejaswini, A. Kumar, I. S. Manochitra, I. S. Chakrapani, and K. Alaskar, "IoT based mobile app for continuous health monitoring of the person," in *Proc. 4th Int. Conf. Emerg. Res. Electron., Comput. Sci. Technol. (ICERECT)*, Dec. 2022, pp. 1–6.
- [66] M. M. Khan, T. M. Alanazi, A. A. Albraikan, and F. A. Almalkim, "IoT-based health monitoring system development and analysis," *Secur. Commun. Netw.*, vol. 2022, no. 1, 2022, Art. no. 9639195.
- [67] M. Yuvaraj, R. Naveen, N. Gokul, S. Gowtham, and S. Ramkumar, "IoT based patient's smart healthcare monitoring and recording using gsm module," in *Proc. E3S Web Conf.*, vol. 399, 2023, p. 4008.
- [68] S. Becker, T. Miron-Shatz, N. Schumacher, J. Krocza, C. Diamantidis, and U.-V. Albrecht, "MHealth 2.0: Experiences, possibilities, and perspectives," *JMIR mHealth uHealth*, vol. 2, no. 2, p. e3328, May 2014.
- [69] C. A. Meier, M. C. Fitzgerald, and J. M. Smith, "EHealth: Extending, enhancing, and evolving health care," *Annu. Rev. Biomed. Eng.*, vol. 15, no. 1, pp. 359–382, Jul. 2013.
- [70] S. L. Ullo and G. R. Sinha, "Advances in smart environment monitoring systems using IoT and sensors," *Sensors*, vol. 20, no. 11, p. 3113, May 2020.
- [71] Y. Cheng, K. Wang, H. Xu, T. Li, Q. Jin, and D. Cui, "Recent developments in sensors for wearable device applications," *Anal. Bioanal. Chem.*, vol. 413, no. 24, pp. 6037–6057, Oct. 2021.
- [72] M. J. McGrath and C. N. Scanaill, Sensor Technologies: Healthcare, Wellness, and Environmental Applications. Cham, Switzerland: Springer, 2013.
- [73] H. Mshali, T. Lemlouma, M. Moloney, and D. Magoni, "A survey on health monitoring systems for health smart homes," *Int. J. Ind. Ergonom.*, vol. 66, pp. 26–56, Jul. 2018.
- [74] J. Fraden and J. King, Handbook of Modern Sensors: Physics, Designs, and Applications, vol. 3. Cham, Switzerland: Springer, 2004.
- [75] M. Javaid, A. Haleem, R. P. Singh, S. Rab, and R. Suman, "Significance of sensors for industry 4.0: Roles, capabilities, and applications," *Sensors Int.*, vol. 2, Jul. 2021, Art. no. 100110.
- [76] J. Vetelino and A. Reghu, *Introduction to Sensors*. Boca Raton, FL, USA: CRC Press, 2017.
- [77] B. Kaur, S. Kumar, and B. K. Kaushik, "Novel wearable optical sensors for vital health monitoring systems—A review," *Biosensors*, vol. 13, no. 2, p. 181, Jan. 2023.
- [78] T. D. da Costa, M. de Fatima Fernandes Vara, C. S. Cristino, T. Z. Zanella, G. N. N. Neto, and P. Nohama, "Breathing monitoring and pattern recognition with wearable sensors," in *Wearable Devices*, N. Nasiri, Ed., Rijeka, Croatia: IntechOpen, 2019, ch. 4, doi: 10.5772/intechopen.85460.
- [79] A. Ruiz-Vargas, S. A. Morris, R. H. Hartley, and J. W. Arkwright, "Optical flow sensor for continuous invasive measurement of blood flow velocity," *J. Biophotonics*, vol. 12, no. 10, Oct. 2019, Art. no. e201900139.
- [80] C. Huang, Y. Gu, J. Chen, A. A. Bahrani, E. G. Abu Jawdeh, H. S. Bada, K. Saatman, G. Yu, and L. Chen, "A wearable fiberless optical sensor for continuous monitoring of cerebral blood flow in mice," *IEEE J. Sel. Topics Quantum Electron.*, vol. 25, no. 1, pp. 1–8, Jan. 2019.
- [81] X. Liu, Y. Gu, C. Huang, M. Zhao, Y. Cheng, E. G. A. Jawdeh, H. S. Bada, L. Chen, and G. Yu, "Simultaneous measurements of tissue blood flow and oxygenation using a wearable fiber-free optical sensor," *J. Biomed. Opt.*, vol. 26, no. 1, Jan. 2021, Art. no. 012705.

- [82] S.-J. Chen, D. C. Hovde, K. A. Peterson, and A. W. Marshall, "Fire detection using smoke and gas sensors," *Fire Saf. J.*, vol. 42, no. 8, pp. 507–515, Nov. 2007.
- [83] T. Dean, T. Brice, A. Hartog, E. Kragh, D. Molteni, and K. O'Connell, "Distributed vibration sensing for seismic acquisition," *Lead. Edge*, vol. 35, no. 7, pp. 600–604, Jul. 2016.
- [84] J. Jia, H. Sun, C. Jiang, K. Karila, M. Karjalainen, E. Ahokas, E. Khoramshahi, P. Hu, C. Chen, T. Xue, T. Wang, Y. Chen, and J. Hyyppä, "Review on active and passive remote sensing techniques for road extraction," *Remote Sens.*, vol. 13, no. 21, p. 4235, Oct. 2021.
- [85] M. Javaid, A. Haleem, S. Rab, R. Pratap Singh, and R. Suman, "Sensors for daily life: A review," *Sensors Int.*, vol. 2, Jun. 2021, Art. no. 100121.
- [86] J. Baranwal, B. Barse, G. Gatto, G. Broncova, and A. Kumar, "Electrochemical sensors and their applications: A review," *Chemosensors*, vol. 10, no. 9, p. 363, 2022.
- [87] Y. Jiang, S. Yin, J. Dong, and O. Kaynak, "A review on soft sensors for monitoring, control, and optimization of industrial processes," *IEEE Sensors J.*, vol. 21, no. 11, pp. 12868–12881, Jun. 2021.
- [88] C. Wang, T. He, H. Zhou, Z. Zhang, and C. Lee, "Artificial intelligence enhanced sensors-enabling technologies to next-generation healthcare and biomedical platform," *Bioelectron. Med.*, vol. 9, no. 1, p. 17, Aug. 2023.
- [89] N. Yang, Z. Wang, R. Gravina, and G. Fortino, "A survey of open body sensor networks: Applications and challenges," in *Proc. 14th IEEE Annu. Consum. Commun. Netw. Conf. (CCNC)*, Jan. 2017, pp. 65–70.
- [90] M. Kompara, S. Kumari, and M. Hölbl, "Analysis and improvement of a secure key management protocol for e-health applications," *Comput. Electr. Eng.*, vol. 73, pp. 97–113, Jan. 2019.
- [91] X. Lai, Q. Liu, X. Wei, W. Wang, G. Zhou, and G. Han, "A survey of body sensor networks," *Sensors*, vol. 13, no. 5, pp. 5406–5447, Apr. 2013.
- [92] C. Crean, C. Mcgeouge, and R. O'kennedy, "Wearable biosensors for medical applications," in *Biosensors for Medical Applications*. Amsterdam, The Netherlands: Elsevier, 2012, pp. 301–330.
- [93] B. M. Mahmmod, S. H. Abdulhussain, M. A. Naser, M. Alsabah, A. Hussain, and D. Al-Jumeily, "3D object recognition using fast overlapped block processing technique," *Sensors*, vol. 22, no. 23, p. 9209, Nov. 2022. [Online]. Available: https://www.mdpi.com/1424-8220/22/23/9209
- [94] B. M. Mahmmod, S. H. Abdulhussain, T. Suk, M. Alsabah, and A. Hussain, "Accelerated and improved stabilization for high order moments of Racah polynomials," *IEEE Access*, vol. 11, pp. 110502–110521, 2023. [Online]. Available: https://ieeexplore. ieee.org/document/10271275/
- [95] K. A. AL-Utaibi, S. H. Abdulhussain, B. M. Mahmmod, M. A. Naser, M. Alsabah, and S. M. Sait, "Reliable recurrence algorithm for high-order Krawtchouk polynomials," *Entropy*, vol. 23, no. 9, p. 1162, Sep. 2021. [Online]. Available: https://www.mdpi.com/1099-4300/23/9/1162
- [96] S. H. Abdulhussain, B. M. Mahmmod, A. AlGhadhban, and J. Flusser, "Face recognition algorithm based on fast computation of orthogonal moments," *Mathematics*, vol. 10, no. 15, p. 2721, Aug. 2022. [Online]. Available: https://www.mdpi.com/2227-7390/10/15/2721
- [97] S. H. Abdulhussain, B. M. Mahmmod, T. Baker, and D. Al-Jumeily, "Fast and accurate computation of high-order Tchebichef polynomials," *Concurrency Comput., Pract. Exper.*, vol. 34, no. 27, pp. 1–10, Dec. 2022. [Online]. Available: https://onlinelibrary. wiley.com/doi/10.1002/cpe.7311
- [98] C. S. Evangeline and A. Lenin, "Human health monitoring using wearable sensor," *Sensor Rev.*, vol. 39, no. 3, pp. 364–376, May 2019.
- [99] N. Gupta, S. K. Gupta, R. K. Pathak, V. Jain, P. Rashidi, and J. S. Suri, "Human activity recognition in artificial intelligence framework: A narrative review," *Artif. Intell. Rev.*, vol. 55, no. 6, pp. 4755–4808, Aug. 2022.
- [100] S. Zhang, Y. Li, S. Zhang, F. Shahabi, S. Xia, Y. Deng, and N. Alshurafa, "Deep learning in human activity recognition with wearable sensors: A review on advances," *Sensors*, vol. 22, no. 4, p. 1476, Feb. 2022.
- [101] T. G. Stavropoulos, A. Papastergiou, L. Mpaltadoros, S. Nikolopoulos, and I. Kompatsiaris, "IoT wearable sensors and devices in elderly care: A literature review," *Sensors*, vol. 20, no. 10, p. 2826, May 2020.
- [102] D. Metcalf, S. T. J. Milliard, M. Gomez, and M. Schwartz, "Wearables and the Internet of Things for health: Wearable, interconnected devices promise more efficient and comprehensive health care," *IEEE Pulse*, vol. 7, no. 5, pp. 35–39, Sep. 2016.

- [103] S. Seneviratne, Y. Hu, T. Nguyen, G. Lan, S. Khalifa, K. Thilakarathna, M. Hassan, and A. Seneviratne, "A survey of wearable devices and challenges," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 4, pp. 2573–2620, 4th Quart., 2017.
- [104] Q. H. Abbasi, H. Heidari, and A. Alomainy, "Wearable wireless devices," *Appl. Sci.*, vol. 9, no. 13, p. 2643, Jun. 2019.
- [105] F. Hudson and C. Clark, "Wearables and medical interoperability: The evolving frontier," *Computer*, vol. 51, no. 9, pp. 86–90, Sep. 2018.
- [106] T. Liang and Y. J. Yuan, "Wearable medical monitoring systems based on wireless networks: A review," *IEEE Sensors J.*, vol. 16, no. 23, pp. 8186–8199, Dec. 2016.
- [107] H. R. Abuzeid, A. F. Abdelaal, S. Elsharkawy, and G. A. Ali, "Basic principles and applications of biological sensors technology," in *Handbook of Nanosensors: Materials and Technological Applications*. Cham, Switzerland: Springer, 2023, pp. 1–45.
- [108] M. Magno, G. A. Salvatore, P. Jokic, and L. Benini, "Self-sustainable smart ring for long-term monitoring of blood oxygenation," *IEEE Access*, vol. 7, pp. 115400–115408, 2019.
- [109] P. Mayer, M. Magno, and L. Benini, "A low power and smart power unit for kinetic self-sustainable wearable devices," in *Proc. 27th IEEE Int. Conf. Electron., Circuits Syst. (ICECS)*, Nov. 2020, pp. 1–4.
- [110] Q. Zhang, T. Jin, J. Cai, L. Xu, T. He, T. Wang, Y. Tian, L. Li, Y. Peng, and C. Lee, "Wearable triboelectric sensors enabled gait analysis and waist motion capture for IoT-based smart healthcare applications," *Adv. Sci.*, vol. 9, no. 4, Feb. 2022, Art. no. 2103694.
- [111] J. Wan, M. A. A. H. Al-awlaqi, M. Li, M. O'Grady, X. Gu, J. Wang, and N. Cao, "Wearable IoT enabled real-time health monitoring system," *EURASIP J. Wireless Commun. Netw.*, vol. 2018, no. 1, pp. 1–10, Dec. 2018.
- [112] S. Majumder, T. Mondal, and M. Deen, "Wearable sensors for remote health monitoring," *Sensors*, vol. 17, no. 1, p. 130, Jan. 2017.
- [113] J. Heikenfeld, A. Jajack, J. Rogers, P. Gutruf, L. Tian, T. Pan, R. Li, M. Khine, J. Kim, J. Wang, and J. Kim, "Wearable sensors: Modalities, challenges, and prospects," *Lab Chip*, vol. 18, no. 2, pp. 217–248, 2018.
- [114] A. E. Alattar and S. Mohsen, "A survey on smart wearable devices for healthcare applications," *Wireless Pers. Commun.*, vol. 132, no. 1, pp. 775–783, Sep. 2023.
- [115] A. Pantelopoulos and N. Bourbakis, "A survey on wearable biosensor systems for health monitoring," in *Proc. 30th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2008, pp. 4887–4890.
- [116] A. Pantelopoulos and N. G. Bourbakis, "A survey on wearable sensorbased systems for health monitoring and prognosis," *IEEE Trans. Syst.*, *Man, Cybern., C, Appl. Rev.*, vol. 40, no. 1, pp. 1–12, Jan. 2010.
- [117] V. Vijayan, J. P. Connolly, J. Condell, N. McKelvey, and P. Gardiner, "Review of wearable devices and data collection considerations for connected health," *Sensors*, vol. 21, no. 16, p. 5589, Aug. 2021.
- [118] P. Cedillo, C. Sanchez, K. Campos, and A. Bermeo, "A systematic literature review on devices and systems for ambient assisted living: Solutions and trends from different user perspectives," in *Proc. Int. Conf. eDemocracy eGovernment (ICEDEG)*, Apr. 2018, pp. 59–66.
- [119] H. Teymourian, M. Parrilla, J. R. Sempionatto, N. F. Montiel, A. Barfidokht, R. Van Echelpoel, K. De Wael, and J. Wang, "Wearable electrochemical sensors for the monitoring and screening of drugs," ACS Sensors, vol. 5, no. 9, pp. 2679–2700, Sep. 2020.
- [120] D. S. Bhatti, S. Saleem, A. Imran, Z. Iqbal, A. Alzahrani, H. Kim, and K.-I. Kim, "A survey on wireless wearable body area networks: A perspective of technology and economy," *Sensors*, vol. 22, no. 20, p. 7722, Oct. 2022.
- [121] J. Li et al., "Thin, soft, wearable system for continuous wireless monitoring of artery blood pressure," *Nature Commun.*, vol. 14, no. 1, p. 5009, Aug. 2023.
- [122] Y. Mao, C. You, J. Zhang, K. Huang, and K. B. Letaief, "A survey on mobile edge computing: The communication perspective," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 4, pp. 2322–2358, 4th Quart., 2017.
- [123] V. S. Naresh, S. S. Pericherla, P. Sita Rama Murty, and S. Reddi, "Internet of Things in healthcare: Architecture, applications, challenges, and solutions," *Comput. Syst. Sci. Eng.*, vol. 35, no. 6, pp. 411–421, 2020.
- [124] B.-S. Kim, K. Kim, and K.-I. Kim, "A survey on mobility support in wireless body area networks," *Sensors*, vol. 17, no. 4, p. 797, Apr. 2017.
- [125] J. V. Ananthi and P. S. H. Jose, "A perspective review of security challenges in body area networks for healthcare applications," *Int. J. Wireless Inf. Netw.*, vol. 28, no. 4, pp. 451–466, Dec. 2021.

- [126] R. A. Khan and A.-S.-K. Pathan, "The state-of-the-art wireless body area sensor networks: A survey," *Int. J. Distrib. Sensor Netw.*, vol. 14, no. 4, Apr. 2018, Art. no. 155014771876899.
- [127] S. Yellampalli, Wireless Sensor Networks: Design, Deployment and Applications. Norderstedt, Germany: Books on Demand, 2021.
- [128] M. Yaghoubi, K. Ahmed, and Y. Miao, "Wireless body area network (WBAN): A survey on architecture, technologies, energy consumption, and security challenges," *J. Sensor Actuator Netw.*, vol. 11, no. 4, p. 67, Oct. 2022.
- [129] S. Movassaghi, M. Abolhasan, J. Lipman, D. Smith, and A. Jamalipour, "Wireless body area networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 3, pp. 1658–1686, 3rd Quart., 2014.
- [130] A. Al-Fuqaha, M. Guizani, M. Mohammadi, M. Aledhari, and M. Ayyash, "Internet of Things: A survey on enabling technologies, protocols, and applications," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2347–2376, 4th Quart., 2015.
- [131] M. Bansal and B. Gandhi, "IoT & big data in smart healthcare (ECG monitoring)," in *Proc. Int. Conf. Mach. Learn., Big Data, Cloud Parallel Comput. (COMITCon)*, Feb. 2019, pp. 390–396.
- [132] M. Kuzlu, M. Pipattanasomporn, and S. Rahman, "Review of communication technologies for smart homes/building applications," in *Proc. IEEE Innovat. Smart Grid Technol.-Asia (ISGT)*, May 2015, pp. 1–6.
- [133] D. Koutras, G. Stergiopoulos, T. Dasaklis, P. Kotzanikolaou, D. Glynos, and C. Douligeris, "Security in IoMT communications: A survey," *Sensors*, vol. 20, no. 17, p. 4828, Aug. 2020.
- [134] G. Calcagnini, E. Mattei, F. Censi, M. Triventi, R. Lo Sterzo, E. Marchetta, and P. Bartolini, "Electromagnetic compatibility of WiFi technology with life-supporting medical devices," in *Proc. World Congr. Med. Phys. Biomed. Eng.*, Munich, Germany. Cham, Switzerland: Springer, 2009, pp. 616–619.
- [135] F. Wu, T. Wu, and M. R. Yuce, "An Internet-of-Things (IoT) network system for connected safety and health monitoring applications," *Sensors*, vol. 19, no. 1, p. 21, Dec. 2018.
- [136] C.-Y. Chen, M.-H. Cheng, M. Cheng, and C.-F. Yang, "Using iBeacon components to design and fabricate low-energy and simple indoor positioning method," *Sensors Mater.*, vol. 35, no. 3, p. 703, 2023.
- [137] J. Ye, H. Gharavi, and B. Hu, "Fast beam discovery and adaptive transmission under frequency selective attenuations in sub-terahertz bands," *IEEE Trans. Signal Process.*, vol. 71, pp. 727–740, 2023.
- [138] C. W. Badenhop, S. R. Graham, B. W. Ramsey, B. E. Mullins, and L. O. Mailloux, "The Z-wave routing protocol and its security implications," *Comput. Secur.*, vol. 68, pp. 112–129, Jul. 2017.
- [139] H. Cao, V. Leung, C. Chow, and H. Chan, "Enabling technologies for wireless body area networks: A survey and outlook," *IEEE Commun. Mag.*, vol. 47, no. 12, pp. 84–93, Dec. 2009.
- [140] A. Yeole, D. R. Kalbande, and A. Sharma, "Security of 6LoWPAN IoT networks in hospitals for medical data exchange," *Proc. Comput. Sci.*, vol. 152, pp. 212–221, Jan. 2019.
- [141] A. Verma and V. Ranga, "Security of RPL based 6LoWPAN networks in the Internet of Things: A review," *IEEE Sensors J.*, vol. 20, no. 11, pp. 5666–5690, Jun. 2020.
- [142] X. Li, G. Chen, B. Zhao, and X. Liang, "A kind of intelligent lighting control system using the EnOcean network," in *Proc. Int. Conf. Comput.*, *Inf. Telecommun. Syst. (CITS)*, Jul. 2014, pp. 1–5.
- [143] T. Borgohain, U. Kumar, and S. Sanyal, "Survey of security and privacy issues of Internet of Things," 2015, arXiv:1501.02211.
- [144] D. M. Mendez, I. Papapanagiotou, and B. Yang, "Internet of Things: Survey on security and privacy," 2017, arXiv:1707.01879.
- [145] S. Adibi, "Link technologies and BlackBerry mobile health (mHealth) solutions: A review," *IEEE Trans. Inf. Technol. Biomed.*, vol. 16, no. 4, pp. 586–597, Jul. 2012.
- [146] J. P. Shanmuga Sundaram, W. Du, and Z. Zhao, "A survey on LoRa networking: Research problems, current solutions, and open issues," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 1, pp. 371–388, 1st Quart., 2020.
- [147] W.-W. Yu and C.-Y. Fang, "The role of near-field communication mobile payments in sustainable restaurant operations: A restaurateur's perspective," *Sustainability*, vol. 15, no. 16, p. 12471, Aug. 2023.
- [148] S. H. Ahammad, D. Nagajyothi, P. P. Priya, D. Sreelakshmi, B. R. Altahan, L. K. Smirani, M. A. Hossain, and A. N. Z. Rashed, "Defected ground-structured symmetric circular ring antenna for near-field scanning plasmonic applications," *Plasmonics*, vol. 18, no. 2, pp. 541–549, Apr. 2023.

- [149] A. S. Albahri, J. K. Alwan, Z. K. Taha, S. F. Ismail, R. A. Hamid, A. A. Zaidan, O. S. Albahri, B. B. Zaidan, A. H. Alamoodi, and M. A. Alsalem, "IoT-based telemedicine for disease prevention and health promotion: State-of-the-art," *J. Netw. Comput. Appl.*, vol. 173, Jan. 2021, Art. no. 102873.
- [150] R. Altawy and A. M. Youssef, "Security tradeoffs in cyber physical systems: A case study survey on implantable medical devices," *IEEE Access*, vol. 4, pp. 959–979, 2016.
- [151] H. Fotouhi, A. Causevic, K. Lundqvist, and M. Björkman, "Communication and security in health monitoring systems—A review," in *Proc. IEEE* 40th Annu. Comput. Softw. Appl. Conf. (COMPSAC), vol. 1, Jun. 2016, pp. 545–554.
- [152] H. Javdani and H. Kashanian, "Internet of Things in medical applications with a service-oriented and security approach: A survey," *Health Technol.*, vol. 8, nos. 1–2, pp. 39–50, May 2018.
- [153] L. M. Dang, M. J. Piran, D. Han, K. Min, and H. Moon, "A survey on Internet of Things and cloud computing for healthcare," *Electronics*, vol. 8, no. 7, p. 768, Jul. 2019.
- [154] M. T. Arefin, M. H. Ali, and A. K. M. F. Haque, "Wireless body area network: An overview and various applications," J. Comput. Commun., vol. 5, no. 7, pp. 53–64, 2017.
- [155] A. C. W. Wong, D. McDonagh, O. Omeni, C. Nunn, M. Hernandez-Silveira, and A. J. Burdett, "Sensium: An ultra-lowpower wireless body sensor network platform: Design & application challenges," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Sep. 2009, pp. 6576–6579.
- [156] M. Joshi, S. Archer, A. Morbi, S. Arora, R. Kwasnicki, H. Ashrafian, S. Khan, G. Cooke, and A. Darzi, "Short-term wearable sensors for in-hospital medical and surgical patients: Mixed methods analysis of patient perspectives," *JMIR Perioperative Med.*, vol. 4, no. 1, Apr. 2021, Art. no. e18836.
- [157] X. Yu, X. Xia, and X. Chen, "Design and application of RuBee-based telemedicine data acquisition system," in *Proc. 10th IEEE/ACIS Int. Conf. Comput. Inf. Sci.*, May 2011, pp. 365–370.
- [158] M. Zareei, A. Zarei, R. Budiarto, and Mohd. A. Omar, "A comparative study of short range wireless sensor network on high density networks," in *Proc. 17th Asia–Pacific Conf. Commun.*, Oct. 2011, pp. 247–252.
- [159] H. Sun, Z. Zhang, R. Q. Hu, and Y. Qian, "Wearable communications in 5G: Challenges and enabling technologies," *IEEE Veh. Technol. Mag.*, vol. 13, no. 3, pp. 100–109, Sep. 2018.
- [160] M. Armbrust, A. Fox, R. Griffith, A. D. Joseph, R. Katz, A. Konwinski, G. Lee, D. Patterson, A. Rabkin, I. Stoica, and M. Zaharia, "A view of cloud computing," *Commun. ACM*, vol. 53, no. 4, pp. 50–58, 2010.
- [161] B. B. P. Rao, P. Saluia, N. Sharma, A. Mittal, and S. V. Sharma, "Cloud computing for Internet of Things & sensing based applications," in *Proc.* 6th Int. Conf. Sens. Technol. (ICST), Dec. 2012, pp. 374–380.
- [162] G. Ramesh, J. Logeshwaran, and V. Aravindarajan, "A secured database monitoring method to improve data backup and recovery operations in cloud computing," *BOHR Int. J. Comput. Sci.*, vol. 2, no. 1, pp. 1–7, 2023.
- [163] A. A. Khamis, A. Idris, A. Abdellatif, N. A. Mohd Rom, T. Khamis, M. S. Ab Karim, S. Janasekaran, and R. B. Abd Rashid, "Development and performance evaluation of an IoT-integrated breath analyzer," *Int. J. Environ. Res. Public Health*, vol. 20, no. 2, p. 1319, Jan. 2023.
- [164] D. K. Sharma, D. S. Chakravarthi, A. A. Shaikh, A. A. A. Ahmed, S. Jaiswal, and M. Naved, "The aspect of vast data management problem in healthcare sector and implementation of cloud computing technique," *Mater. Today, Proc.*, vol. 80, pp. 3805–3810, Jan. 2023.
- [165] R. Bryant, R. H. Katz, and E. D. Lazowska, "Big-data computing: Creating revolutionary breakthroughs in commerce, science and society," Tech. Rep., 2008. [Online]. Available: https://cra.org/ccc/wpcontent/uploads/sites/2/2015/05/Big_Data.pdf
- [166] M. Bamiah, S. Brohi, S. Chuprat, and J.-L. A. Manan, "A study on significance of adopting cloud computing paradigm in healthcare sector," in *Proc. Int. Conf. Cloud Comput. Technol., Appl. Manage. (ICCCTAM)*, Dec. 2012, pp. 65–68.
- [167] R. Chauhan and A. Kumar, "Cloud computing for improved healthcare: Techniques, potential and challenges," in *Proc. E-Health Bioeng. Conf.* (*EHB*), Nov. 2013, pp. 1–4.
- [168] N. Sultan, "Making use of cloud computing for healthcare provision: Opportunities and challenges," *Int. J. Inf. Manage.*, vol. 34, no. 2, pp. 177–184, Apr. 2014.
- [169] N. John and S. Shenoy, "Health cloud–healthcare as a service (HaaS)," in Proc. Int. Conf. Adv. Comput., Commun. Informat. (ICACCI), Sep. 2014, pp. 1963–1966.

- [170] S. Oueida, Y. Kotb, M. Aloqaily, Y. Jararweh, and T. Baker, "An edge computing based smart healthcare framework for resource management," *Sensors*, vol. 18, no. 12, p. 4307, Dec. 2018.
- [171] S. Yi, C. Li, and Q. Li, "A survey of fog computing: Concepts, applications and issues," in *Proc. Workshop Mobile Big Data*, 2015, pp. 37–42.
- [172] M. Chiang and T. Zhang, "Fog and IoT: An overview of research opportunities," *IEEE Internet Things J.*, vol. 3, no. 6, pp. 854–864, Dec. 2016.
- [173] F. Al-Doghman, N. Moustafa, I. Khalil, N. Sohrabi, Z. Tari, and A. Y. Zomaya, "AI-enabled secure microservices in edge computing: Opportunities and challenges," *IEEE Trans. Services Comput.*, vol. 16, no. 2, pp. 1485–1504, Mar. 2023.
- [174] T. Kim, S. D. Sathyanarayana, S. Chen, Y. Im, X. Zhang, S. Ha, and C. Joe-Wong, "MoDEMS: Optimizing edge computing migrations for user mobility," *IEEE J. Sel. Areas Commun.*, vol. 41, no. 3, pp. 675–689, Mar. 2023.
- [175] C. Li, J. Wang, S. Wang, and Y. Zhang, "A review of IoT applications in healthcare," *Neurocomputing*, vol. 565, Jan. 2024, Art. no. 127017.
- [176] J. Kharel, H. T. Reda, and S. Y. Shin, "An architecture for smart health monitoring system based," J. Commun., vol. 12, no. 4, pp. 228–233, 2017.
- [177] I. V. Pustokhina, D. A. Pustokhin, D. Gupta, A. Khanna, K. Shankar, and G. N. Nguyen, "An effective training scheme for deep neural network in edge computing enabled Internet of Medical Things (IoMT) systems," *IEEE Access*, vol. 8, pp. 107112–107123, 2020.
- [178] D. Kreutz, F. M. V. Ramos, P. E. Veríssimo, C. E. Rothenberg, S. Azodolmolky, and S. Uhlig, "Software-defined networking: A comprehensive survey," *Proc. IEEE*, vol. 103, no. 1, pp. 14–76, Jan. 2015.
- [179] R. Mijumbi, J. Serrat, J.-L. Gorricho, N. Bouten, F. De Turck, and R. Boutaba, "Network function virtualization: State-of-the-art and research challenges," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 1, pp. 236–262, 1st Quart., 2016.
- [180] Y. Li and M. Chen, "Software-defined network function virtualization: A survey," *IEEE Access*, vol. 3, pp. 2542–2553, 2015.
- [181] X. Huang, R. Yu, J. Kang, Y. He, and Y. Zhang, "Exploring mobile edge computing for 5G-enabled software defined vehicular networks," *IEEE Wireless Commun.*, vol. 24, no. 6, pp. 55–63, Dec. 2017.
- [182] N. Hassan, S. Gillani, E. Ahmed, I. Yaqoob, and M. Imran, "The role of edge computing in Internet of Things," *IEEE Commun. Mag.*, vol. 56, no. 11, pp. 110–115, Nov. 2018.
- [183] E. Peltonen, M. Bennis, M. Capobianco, M. Debbah, A. Ding, F. Gil-Castiñeira, M. Jurmu, T. Karvonen, M. Kelanti, A. Kliks, T. Leppänen, L. Lovén, T. Mikkonen, A. Rao, S. Samarakoon, K. Seppänen, P. Sroka, S. Tarkoma, and T. Yang, "6G white paper on edge intelligence," 2020, arXiv:2004.14850.
- [184] Q.-V. Pham, F. Fang, V. N. Ha, Md. J. Piran, M. Le, L. B. Le, W.-J. Hwang, and Z. Ding, "A survey of multi-access edge computing in 5G and beyond: Fundamentals, technology integration, and state-of-the-art," *IEEE Access*, vol. 8, pp. 116974–117017, 2020.
- [185] A. M. Rahmani, T. N. Gia, B. Negash, A. Anzanpour, I. Azimi, M. Jiang, and P. Liljeberg, "Exploiting smart e-Health gateways at the edge of healthcare Internet-of-Things: A fog computing approach," *Future Gener. Comput. Syst.*, vol. 78, pp. 641–658, Jan. 2018.
- [186] P. Mach and Z. Becvar, "Mobile edge computing: A survey on architecture and computation offloading," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 3, pp. 1628–1656, 3rd Quart., 2017.
- [187] X. Li, X. Huang, C. Li, R. Yu, and L. Shu, "EdgeCare: Leveraging edge computing for collaborative data management in mobile healthcare systems," *IEEE Access*, vol. 7, pp. 22011–22025, 2019.
- [188] A. Awad Abdellatif, A. Emam, C.-F. Chiasserini, A. Mohamed, A. Jaoua, and R. Ward, "Edge-based compression and classification for smart healthcare systems: Concept, implementation and evaluation," *Expert Syst. Appl.*, vol. 117, pp. 1–14, Mar. 2019.
- [189] C.-Y. Chang, K. Alexandris, N. Nikaein, K. Katsalis, and T. Spyropoulos, "MEC architectural implications for LTE/LTE–A networks," in *Proc. Workshop Mobility Evolving Internet Archit.*, Oct. 2016, pp. 13–18.
- [190] S. D. A. Shah, M. A. Gregory, S. Li, and R. D. R. Fontes, "SDN enhanced multi-access edge computing (MEC) for E2E mobility and QoS management," *IEEE Access*, vol. 8, pp. 77459–77469, 2020.
- [191] I. M. Ali and M. I. Salman, "SDN-assisted service placement for the IoT-based systems in multiple edge servers environment," *Iraqi J. Sci.*, vol. 10, pp. 1525–1540, Jan. 2020.

- [192] H. L. Pham, T. H. Tran, and Y. Nakashima, "A secure remote healthcare system for hospital using blockchain smart contract," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, Dec. 2018, pp. 1–6.
- [193] K. N. Griggs, O. Ossipova, C. P. Kohlios, A. N. Baccarini, E. A. Howson, and T. Hayajneh, "Healthcare blockchain system using smart contracts for secure automated remote patient monitoring," *J. Med. Syst.*, vol. 42, no. 7, pp. 1–7, Jul. 2018.
- [194] M. Antwi, A. Adnane, F. Ahmad, R. Hussain, M. Habib ur Rehman, and C. A. Kerrache, "The case of HyperLedger fabric as a blockchain solution for healthcare applications," *Blockchain, Res. Appl.*, vol. 2, no. 1, Mar. 2021, Art. no. 100012.
- [195] J. Brogan, I. Baskaran, and N. Ramachandran, "Authenticating health activity data using distributed ledger technologies," *Comput. Struct. Biotechnol. J.*, vol. 16, pp. 257–266, Jan. 2018.
- [196] A. Kapoor, S. Guha, M. Kanti Das, K. C. Goswami, and R. Yadav, "Digital healthcare: The only solution for better healthcare during COVID-19 pandemic?" *Indian Heart J.*, vol. 72, no. 2, pp. 61–64, Mar. 2020.
- [197] M. Zhang, A. Raghunathan, and N. K. Jha, "Trustworthiness of medical devices and body area networks," *Proc. IEEE*, vol. 102, no. 8, pp. 1174–1188, Aug. 2014.

MUNTADHER ALSABAH received the B.S. degree in electrical engineering from the University of Baghdad, Iraq, in 2007, the M.S. degree in wireless communication engineering from Universiti Putra Malaysia (UPM), Malaysia, in 2014, and the Ph.D. degree in wireless communication from The University of Sheffield, Sheffield, U.K., in 2020. His current research interests include wireless communications, multiple-inputmultiple-output (MIMO) systems, mmWave fre-

quencies, image processing, signal processing, and applying artificial intelligence, and machine learning techniques in wireless communications systems.

HAYA ALASKAR received the M.Sc. degree in applied artificial intelligence

from the University of Exeter, in 2009, and the Ph.D. degree in computer

science from Liverpool John Moores University, in 2014. She is currently an

Assistant Professor with the College of Computer Science and Engineering,

Prince Sattam Bin Abdulaziz University, Saudi Arabia. She has a number

of publications concentrated on using machine learning in various medical

data, such as signals and images. Her research interests include artificial

FAHDAH ALMARSHAD is currently a Professor (Assistant) with Prince Sattam Bin Abdulaziz University, Saudi Arabia. She is interested in multidisciplinary research topics related to computer science. Her research interests include machine learning, cyber security standards, information assurance, security risks, security policies, security culture, and the Internet

intelligence applications and data science.

HALA J. MOHAMMED received the M.Sc. degree in computer engineering from the University of Baghdad, in 2019. She is currently an Assistant Lecturer with the Department of Computer Engineering, University of Baghdad. Her research interests include speech enhancement, network-on-chip, and computer networks.

BASHEERA M. MAHMMOD received the Ph.D. degree in computer and embedded system engineering from UPM, in 2018. She is currently a Lecturer with the Department of Computer Engineering, University of Baghdad. Her research interests include signal processing, speech enhancement, and computer vision.

MARWAH ABDULRAZZAQ NASER received the B.Sc. degree in electronics and communication engineering from the University of Baghdad, Iraq, in 2008, and the M.Sc. degree in wireless communication systems from The University of Sheffield, U.K., in 2016. She is currently an Assistant Lecturer with the University of Baghdad. Her current research interests include wireless communication systems, signal processing, image processing, and mobile networks.

of Things security.

ABIR HUSSAIN (Senior Member, IEEE) is currently a Professor of machine learning with the Department of Computer Science, Liverpool John Moores University. Her research interests include machine learning algorithms and their applications to medical, image and signal processing, and data analysis.

AHLAM H. SHANIN AL-SUDANI received the M.Sc. degree in control and computer engineering from the University of Baghdad, in 2005. She is currently a Lecturer with the Department of Computer Engineering, University of Baghdad. Her research interests include digital signal processing, image computing, and computer vision.

SADIQ H. ABDULHUSSAIN received the B.S. degree in electrical engineering and the M.S. degree in electronics and communication engineering from the University of Baghdad, in 1998 and 2001, respectively, and the Ph.D. degree in computer and embedded system engineering from Universiti Putra Malaysia, in 2018. Since 2005, he has been a Staff Member with the Department of Computer Engineering, Faculty of Engineering, University of Baghdad. His research

interests include computer vision, signal processing, speech and image processing, and communication.